Contents lists available at ScienceDirect



Renewable and Sustainable Energy Reviews

journal homepage: www.elsevier.com/locate/rser

Demand side management in microgrid: A critical review of key issues and recent trends





Dharmaraj Kanakadhurga, Natarajan Prabaharan*

School of Electrical & Electronics Engineering, Department of EEE, SASTRA Deemed University, Thanjavur 613401, Tamilnadu, India

ARTICLE INFO

Keywords: Distributed generation Microgrid Demand side management Electric vehicles Power quality Controllers Soft computing techniques

ABSTRACT

In a deregulated power system, Demand Side Management (DSM) plays a vital role in handling the uncertain renewable power generation and load. The flat load-profile can be obtained using the Demand Response (DR) techniques with the storage elements and proper switching. The increasing penetration of Renewable Energy Sources (RES) and Electric Vehicles (EVs) supports the DR measures which facilitate both the utility and the consumer. The objective of DSM is to minimize the peak demand, electricity cost and emission rate by the effective utilization of storage with RES. This review article mainly focuses on the layers of microgrid, different techniques involved in DSM, mathematical models of DSM, latest optimization techniques and application of storage devices such as battery energy storage system and EVs in DSM. The state of the art of this article lies on the critical analysis related to datascience, advanced metering infrastructure and blockchain technologies which are the uniqueness of this article. The key issues and approaches are examined critically with the existing works to show how DSM implementation can be effectively done in the microgrids to reduce the electricity cost. This article helps the researchers to identify the research gap by gaining knowledge on the implementation of DSM in the microgrid and the factors affecting the DSM implementation. Few recommendations are discussed to provide future directions for researchers who started working in the DSM implementation.

1. Introduction

Over the decades, the electricity demand has increased more rapidly than the power generation. The demand growth could be handled by the Distributed Generation (DG) which is defined as the installation of generating units such as wind, solar, combined heat and power, diesel generator, small hydro and biomass plants at the user end. Among the DGs, the combined heat and power and diesel generator create environmental pollution which could be effectively handled by the Renewable Energy Sources (RES) based DGs [1]. The RES-based DGs have some advantages such as reduced emissions, no operation cost in terms of fuel, loss minimization, voltage profile improvement, and minimum maintenance [2]. Therefore, it may be used in both utility side and consumer premises such as residential, commercial and industrial. The installation cost and its subsequent cost/kWh is greater than the conventional DGs. Also, its usage is limited due to the drawbacks of uncertainty and intermittency which cause supply-demand imbalance. It requires more reserves to support the peak-time power scarcity. The cost associated with the system increases while increasing the additional reserve capacity in the system [3]. Storage units are the alternative solution to handle the uncertain power generation of RES and ensure the complete utilization of RES-based DG [4]. The power generated from RES on the consumer side deregulates the conventional power system structure and facilitates the system to exchange power from consumer to utility [5].

The International Council on Large Electric Systems (CIGRE) defined Microgrid as, "Microgrids are electricity distribution systems containing loads and distributed energy resources, (such as distributed generators, storage devices, or controllable loads) that can be operated in a controlled, coordinated way either while connected to the main power network or while islanded" [6]. Power mismatch between demand and generation is the major concern to be addressed in both grid-connected and islanded microgrids [7]. In a grid-connected microgrid, power exchanges between the microgrid to the utility grid and vice-versa. The power mismatch problem in grid-connected microgrids could be solved either by varying the generation or varying the demand. Another problem associated with the grid system (e.g., outage) can be decreased by increasing any one of the capacities such as generating capacity, storage capacity and reserve capacity [8]. Increasing capacity in the grid system may increase the investment cost. In islanded operation, the microgrid is not connected to the utility grid and the consumption is served from standalone DGs within the microgrid. The power mismatch problem in islanded microgrids could

* Corresponding author. E-mail address: prabaharan.nataraj@gmail.com (N. Prabaharan).

https://doi.org/10.1016/j.rser.2021.111915

Received 12 May 2021; Received in revised form 4 October 2021; Accepted 15 November 2021 Available online 17 December 2021 1364-0321/ \Circ 2021 Elsevier Ltd. All rights reserved.

List of Abbrev	riations
AMI	Advanced Metering Infrastructure
ANN	Artificial Neural Network
ANSI	American National Standards Institute
ASD	Adjustable Speed Drives
BEE	Bureau of Energy Efficiency
BESS	Battery Energy Storage System
BPSO	Binary PSO
CAES	Compressed air energy storage
CFL	Compact Fluorescent Lamp
CHP	Combined Heat and Power
CMP	Capacity Market Programs
CPP	Critical Peak Pricing
DADR	Data Analytical Demand Response
DB	Demand Bidding
DER	Distributed Energy Resources
DG	Distributed Generation
DISCOM	DIStribution COMpanies
DL	Deep Learning
DLC	Direct Load Control
DR	Demand Response
DSM	Demand Side Management
EDRP	Emergency DR Program
EMS	Energy Management System
ESS	Energy Storage System
EV	Electric Vehicle
GA	Genetic Algorithm
GWDO	Genetic Wind driven optimization
GWO	Grey Wolf Optimization
HEMS	Home Energy Management System
HPP	Hybrid power plant
I/Cs	Interruptible or Curtailable services
IBDR	Incentive-based DR
VPP	Virtual Power Plant
IBR	Inclined Block Rate
IEC	International Electrotechnical Commission
IEEE	Institute of Electrical and Electronics Engineers
ILP	Integer Linear Programming
IP	Internet Protocol
IoT	Internet of Things
ISO	Independent System Operator
LAN	Local Area Network
MILP	Mixed Integer Linear Programming
MSEDL	Maharashtra State Electricity Distribution Com-
	pany Limited
OSR	Optimum Stopping Rule
PAR	Peak to Average Ratio
PHEV	Plugged in Hybrid Electric Vehicle
PLC	Power Line Communication
PQ	Power Quality
PSO	Particle Swarm Optimization
PTR	Peak Time Rebate
PV	PhotoVoltaic
RDL	Reinforcement Deep Learning
RES	Renewable Energy Sources

be solved by implementing a proper control scheme to schedule the load, DGs and storage devices [9,10]. The consensus algorithm-based control has been implemented for resource allocation to reduce the

Renewable and Sustainable Energy Reviews 156 (2022) 111915

RHO	Rolling Horizon Optimization
RL	Reinforcement learning
RTP	Real Time Pricing
SCADA	Supervisory Control And Data Acquisition
SCEM	Smart Compact Energy Meter
SERC	State Electricity Regulatory Commission
SGAM	Smart Grid Architecture Model
ToD	Time of Day
ToU	Time of Use
THD	Total Harmonic Distortion (THD)
V2G	Vehicle to Grid
WDO	Wind Driven Optimization
ML	Machine learning

supply and demand gap where the optimal scheduling has been done with the aid of distributed energy storage [11].

The load pattern should be changed for optimum operation of the system which can be obtained by Demand Side Management (DSM). DSM is the energy management technique that is used to modify the load pattern of the consumer. Energy efficiency and Demand Response (DR) are the two different categories of DSM implementation [12]. Energy efficiency includes the replacement of inefficient older equipment with efficient equipment whereas DR deals with pricing schemes that can provide the benefits on shifting the demand from peak-hour to an off-peak hour to reduce the Peak to Average Ratio (PAR). The DR program such as peak clipping, valley filling, load shifting, load growth, strategic conservation and flexible load curve can be utilized for changing the load pattern. It has many advantages such as cost reduction, improved reliability, lower number of blackouts, reduced dependency on fossil fuels and emissions. From the customer's point of view, DSM helps to reduce their electricity bill. On the utility side, it reduces the investment cost associated with the additional capacity. The implementation of RES-based DGs with DSM will get the maximum benefit on the utility side in terms of power balance during the peak demand period [12].

The installed capacity of RES in India is approximately 92550.74 MW which includes solar power, wind power, bio-power and small hydro-power. As per the ministry of new and renewable energy report, wind power has 38683.65 MW, solar power has 38794.07 MW, biopower has 10314.56 MW and small hydro-power has 4758.46 MW [13]. The energy and resource institute in India has published a document which deals with the future planning of RES installation in India. The report specified that the installation should be 175 GW by 2022 out of which 100 GW from solar power, 60 GW from wind energy, 10 GW from small hydro-power, and 5 GW from biomass-based power [14]. The on-site generation of DSM implementation has become popular in India because of the installation of solar and other renewable energy resources in the consumer end for various applications [15]. Later, the DSM has been implemented for improving energy efficiency by replacing the existing load with the efficient one and encouraging the adoption of energy conservation measures.

The impact of grid integration in microgrids with Distributed Energy Resources (DER), protection schemes, Power Quality (PQ) [16] and stability of the microgrid have been critically reviewed in [17,18]. An exclusive review on central controllers [19] and protection schemes along with control techniques has been discussed in [20]. But the DSM implementation in microgrid has not been critically reviewed in the above-mentioned review articles. The policies related to DSM implementations have been discussed for various countries such as Kuwait [21], China [22] and UK [23]. DR techniques in residential buildings and their control techniques, optimization method and tariff structures have been studied in [24]. The business model of DSM for different segments of the electricity market such as operation, generation, transmission and distribution, retailing and load segments have been reviewed in [25]. The different DSM implementations, tariff structure and the impact of DSM on reliability and security have been reviewed in [26]. The critical assessment on the peak-load shifting for different kinds of consumers such as residential, industrial, campus and islanded microgrids has been done in [27] and also the importance of communication infrastructure has been insisted. DSM implementation in microgrids has been discussed in the above-mentioned review articles.

In earlier days, DSM techniques were implemented on energy conservation and energy efficiency due to lack of on-site generation, communication technologies including Advanced Metering Infrastructure (AMI), data storage & handling techniques and soft computing techniques. The AMI with secure IoT has been developed for DSM implementation in [28]. Recently, the Reinforcement Learning (RL) [29], Deep Learning (DL) [30] and Machine Learning (ML) [31] models have been suggested for enhancing either the load or source forecasting to achieve energy supply-demand balance. Blockchain technology has been suggested for secure data storage and data handling with more privacy in [32]. Thermostatically controlled devices with DSM implementation have been given more importance than electrically controlled devices due to their behavior of decreasing the energy demand. The energy trading with DSM implementation among consumers in real-time frameworks has been suggested for community-based microgrids to reduce the electricity cost. For real-time implementation, sensor network applications have been considered as the major requirement for optimizing energy usage in smart cities and intelligent buildings.

Based on the above discussion, most of the review articles contemplate either only the microgrid control techniques or types of DSM techniques. For a better understanding of DSM implementation in the microgrid, it is necessary to study the microgrid structure including components and controllers. Also, it is imperative to understand the impact of DSM implementation in the microgrid based on the performance parameters which is discussed in Section 6. The performance parameters are the PQ, reliability and stability of the microgrid. Therefore, this article consolidates all the components involved in the implementation of DSM in the microgrid. The available literature provides knowledge about economic performance, environmental, operational and reliability issues in the microgrid without co-relation of each other. In this case, there is a possibility to degrade the microgrid performance if any one of the parameters is considered alone. This review article tries to showcase the joint investigation on the impacts of DSM in the above-mentioned performance parameters which are the uniqueness of this article. Also, this review article discusses the implementation of data science in DSM such as big data, blockchain, Internet of Things (IoT) and ML etc. As a result, this review article provides in-depth and cutting-edge knowledge on the implementation of DSM in the microgrid. Also, it helps the researcher to identify where the DSM could be implemented and how the implementation affects the performance parameters in the microgrid.

The remaining sections of this article as follows: Section 2 deals with the architecture model, major components of the microgrid and the contribution of each layer of the microgrid for DSM implementation. Section 3 elaborates the different DSM implementations and comparative study based on their application, physical components, optimization methods used. Also, this section discusses the utilization of advanced soft computing techniques that makes the DSM implementation simple and accurate. The major physical components in the microgrid that supports the DSM are RES and energy storage devices. So, Section 4 describes the contribution of batteries and EV for the implementation of DSM in the microgrid. Section 5 deals the advanced technologies in terms of metering, data science and communication in the application of DSM in microgrids. Section 6 deals the performance parameters for the implementation of DSM in the microgrid. Finally, the article ends with the conclusion.

2. Microgrid

The microgrid can be defined as a network consisting of loads and DER either connected to the grid known as "grid-connected mode" or isolated from the grid known as "islanded mode". The requirements of a microgrid are (i) distinct from the network, (ii) local control of the energy resources instead of centralized control from the remote control center, and (iii) either grid-connected or islanded mode of operation. To satisfy the requirements, the microgrid needs the appropriate control and maintenance which are monitored by the control centers [33]. The control measure may vary with respect to the grid type. The microgrid is classified based on the application in [33] and size in [34]. Based on the application, the microgrid can be classified as campus/institutional microgrids, military microgrids, residential, remote and rural microgrids. Based on the size of the microgrid, it is classified as nanogrid and picogrid. Nanogrid is one of the types of microgrid consists of the DG and industrial consumers whereas picogrid consists of the small-scale DG and residential consumers [34].

The Smart Grid Architecture Model (SGAM) has been discussed in [35]. The architecture model has been illustrated in Fig. 1. The component layer includes the physical components of the grid such as DGs, storage devices, smart meters, sensors and actuators. The communication layer includes the network and protocols that are connected with other layers to share the data from the information layer. The function layer describes the logical functions and applications of the entire grid. The Business layer handles the business model of the electricity market and its regulatory framework [36].

The five layers of smart grid (business, function, information, communication, and component layer) have been extracted from eight stacks of the grid-wise architectural council in [37]. Among the eight stacks, economic /regulatory policy and business objectives have come under business layer. The function layer has business procedures and context whereas the information layer has data/information that could be communicated between the different systems. The communication mode and infrastructure have been listed under the communication layer whereas the component layer has been included the components required for the grid operation and connectivity. The SGAM in European projects has been critically reviewed in [38]. The wide range of applications, benefits of SGAM and challenges on the implementation of SGAM have been critically reviewed in [38]. It has also discussed the points to be considered for future research in terms of development tools and their implementation/testing.

The architecture of a microgrid consists of five layers such as the component layer, communication layer, information layer, function layer and business layer. The function of each layer has been explained in the subsequent sections. Section 2.1 describes the function of component layer, Section 2.2 discusses the function of communication layer, Section 2.3 discusses the function of both the information layer and the function layer and Section 2.4 discusses the function of the business layer.

2.1. Component layer

The component layer (physical layer) contains the physical components such as DG, storage unit, power electronic converters and loads. A hybrid energy system consists of the combination of the gridconnected DG, storage and load which may be configured as series, parallel and switched [39]. The different DGs used in the residential sector are solar PhotoVoltaic (PV) panels, small wind turbines, naturalgas-fired fuel cells, and diesel generators. The different DGs used in both the commercial and industrial sectors are combined heat and power systems, solar photovoltaic panels, wind, hydropower, biomass combustion or co-firing, municipal solid waste incineration, fuel cells, diesel generators [40]. The utilization of the loads in the microgrid can be traditional or smart. Smart loads are flexible to adopt any advanced technique for implementing the DR whereas the traditional loads can measure only energy efficiency.



Fig. 1. Smart grid architecture model.

2.1.1. Distributed generation

DG is the small-scale generation of electricity at the consumer end. DG technologies consist of modular generators (e.g. diesel generators) and RES. Generally, DGs can provide electrical energy at a low cost or even without cost at higher reliability and security with fewer environmental impacts [41]. Instead of using few large-scale generating stations located far away from load centers, the numerous DG systems may employ as small plants. It possess various potential benefits such as reduced transmission losses at the distribution end and minimum operation cost.

Photovoltaic systems. PV systems convert directly the solar energy into electricity [42]. In [43], PV has been used for the DSM implementation in different types of consumers (residential, industrial and commercial) to minimize the electricity cost. In [44], prosumer-based energy management with the PV power has been implemented with three different tariff structures: Real Time Pricing (RTP), feed-in tariffs and net purchase/ net sale. It was concluded that the net purchase/ net sale was the most economical tariff structure and RTP was the second option. In [45], a Home Energy Management System (HEMS) was developed for a consumer with PV power generation. The efficiency of the developed model has been proved by comparing their electricity cost and peak demand with the traditional consumer and HEMS without PV installation. Energy trading among the neighbors has been suggested in [46,47] for minimization of electricity cost and peak demand. In [46], a multi-agent model has been utilized to share the PV and battery powers among the neighboring consumers. In [47], shared energy storage was developed with the implementation of incentive-based DR and obtained the electricity cost minimization. For a residential consumer, the implementation of DSM with the combination of PV, Battery Energy Storage System (BESS) and EV has reduced the peak demand up to 96% [48].

Wind energy system. Wind power plants convert the kinetic energy of the wind into electrical energy. Distributed wind energy system ranges from a 5 kW turbine at a home to few megawatt turbines at the industrial premises [49]. Variable-speed wind turbines can extract the maximum power at a wide range of wind speeds. The dynamic modeling and control of the small wind-turbine via furling dynamics at high wind-speed maintains the steady-state power. In [50], the hill climbing control has been used to obtain the maximum annual energy capture. In [51], DSM was used for the integration of wind power with the grid to manage the intermittency of wind power generation, minimize the peak-load and avoids the reliability problems. The minimization of operating cost and emission rate was obtained with the implementation of DR program and wind power along with solar power in [52]. DR program with Plugged-in EV (PHEV) and wind power have been implemented for cost savings of the consumer [53]. The electricity cost for the energy consumption of 50 users have been compared with and without DSM implementation where the cost saving for the users have been obtained by DSM implementation [54]. The maximization of wind energy utilization and minimization of cost and carbon emission have been obtained by the coordination of DSM, wind power and BESS [55]. In [56], the energy management of a hybrid system with solar, wind and BESS has been utilized for improving the reliability of the microgrid.

2.1.2. Loads

An electrical load is an appliance or equipment that consumes the electrical energy [57]. It plays a vital role in DSM implementation for load management to modify the load pattern [58]. Electrical appliances, types of equipment, and types of machinery in residential, commercial, and industrial consumers are the different possible loads in a microgrid system [59].

Industrial consumers are the large-scale energy consumers with advanced measurement, control, and communication infrastructures that possess the various advantages such as fast and accurate scheduling [60]. In absence of DG, the modification of industrial consumers load pattern by proper scheduling for minimizing the peak demand [59]. As most of the loads in production industry are uninterruptible. But the interruption in any one of the load may affect the subsequent processes which it requires the technical and financial barriers to be handled along with the DR implementation [60].

Residential consumers are the second largest consumers after industrial consumers. The most common categories are controllable, interruptible, and uninterruptible. Controllable, interruptible, and uninterruptible appliances have been scheduled for minimization of electricity cost, peak load, and PAR. Recently, many research works have focused on the residential consumer along with the implementation of EV [61]. The classification of smart residential appliances for DSM implementation along with tariff has been tabulated in Table 1. From Table 1, it is understood that the classification of different appliances in each category of the residential consumers to the effective DSM implementation.

In [62], the price bound has been communicated with commercial consumer to modify the load pattern for minimizing the electricity cost.

Ref No	Classification of residential	appliances					Tariff
	Controllable	Uninterruptible controllable	Interruptible controllable	Thermostatically controllable	Uncontrollable	Continuous operation	_
[61]	Iron box, Induction motor, Electric water heater, Refrigerator, Compact Fluorescent Lamp (CFL),light emitting diode lamp	Ceiling fan, Grinder, Tube light, Television, Air conditioner, Computer, Laptop, Mobile charger	Washing machine				RTP
[64]	Washing machine, Television, Air conditioner, Lighting						Optimal bidding strategy
[65]	Air conditioner, Oven, Clothes washer, Lighting, Television, Refrigerator						Time of Use (ToU)
[66]	Dish washer, Washing machine, Lighting, Oven, Sprinkler				Stove, Television, Laptop, Iron box, Coffee machine, Shaver, Game consol	Refrigerator, Outdoor lighting	Fixed tariff, Consumption based tariff, Rewarding tariff
[67]	Washing machine, Clothes dryer			Air conditioner, Water heater, Refrigerator, Water dispenser	Water pump, Oven, Dish washer, Vacuum pump		RTP
[68]	Dish washer, Plug-in EV, Electric water heater, Air conditioner Electric rice cooker						ToU
[69]	Battery, EV (Power Shiftable)		Washing machine, Dish washer, Clothes dryer, Automatic defrost (Time shiftable appliances)	Electric water heater, Refrigerator, Air conditioner			ToU, Feed in tariff
[70]	Washing machine, EV, Dish washer, Cloth dryer				Refrigerator, Lighting, Microwave, Television, Vacuum cleaner, Computer, Laptop		RTP
[71]	Washing machine, EV, Spin dryers, Dish washers, Cooker hub, Cooker oven Microwave, Computer, Vacuum cleaner, Laptop			Heating, Ventilation, Air conditioner		Refrigerator	ToU

In [63], automated DR has been analyzed for both industrial and commercial consumers. The peak load has been minimized and reliability was improved by implementing incentive-based DR for industrial and commercial consumers in Delhi.

2.2. Communication layer

The communication layer is responsible for information exchange between the utility and consumers. It supports the energy management and stability of the system [72]. Wide area network [73], field area network, neighborhood area network [73,74] and Local Area Network (LAN) [75] are the communication protocols used in the microgrid. LAN consists of building area network, industrial area network and home area network. In these types of wireless networks, the protocols such as Wireless Fidelity, home plug, digital subscriber line, ZigBee, Dash7 have been chosen based on standards of the Institute of Electrical and Electronics Engineers (IEEE) 802.15.4/Internet Protocol (IP), International Electrotechnical Commission (IEC) 61850, International Organization for Standardization/IEC 18000-7, IEEE 802.15.1, IEEE 802.11 (Wireless Fidelity) and IEEE 802.16 (Worldwide interoperability for microwave access) [76]. In [77], wireless communication has been used for DSM with microcontroller-based control. DSM has been implemented and the flat load profile was obtained without any storage devices. For an urban environment, mobile communication has been suggested for DSM in [78]. It has been proved that a long-range wide area network architecture of mobile communication with DSM implementation for minimization of PAR.

2.3. Intelligence layer

An Intelligence layer is the combination of information and function layers. It monitors the control of physical layer components. The control can be either centralized or decentralized. In the centralized control, the resource allocation has been done by Independent System Operator (ISO) [43]. The demand and DG of a consumer is communicated to the ISO periodically and the ISO manages the supply-demand balance. In decentralized control, the control of load allocation lies within the individual consumer. The consumer can make the scheduling based on the price of electricity on a particular period, storage availability and DG availability. Then, the consumer communicates the demand to the central utility and the utility acts based on the requirement from the consumer [79]. If the demand level quoted by the consumer is very high, the utility will impose a fare for that excessive power consumption. In intermediate control, the individual consumers are given rights to control their demand. If the demand exceeds the threshold, the central controller can keep it within the limit. The different controllers for microgrids [80] are droop control [81], V-f control [82], P-Q sharing [83], Energy Management System (EMS) [84] and several other miscellaneous functions. The energy management in a microgrid with DG is handled by DSM. EMS is classified into two types such as Direct Load Control (DLC) and indirect load control [84] which will be further explained in Sections 3.1 and 3.2.

2.3.1. Centralized control

In centralized control, an aggregator or control center maintains the load pattern of the consumer which does not consider the consumer satisfaction level [85]. Fig. 2 represents the centralized control of microgrid architecture. From Fig. 2, it is evident that the microgrid consists of DGs such as solar PV, wind & diesel generator, battery, EV and load (consumer with different appliances and smart meter). Each microgrid is connected to the grid as well as the other microgrids through a centralized control center. The centralized control center monitors and controls the power generation, power demand and powersharing among the microgrids. The consumer response for changing the load pattern could be improved while implementing the ToU, RTP and Critical Peak Pricing (CPP) tariffs in a centralized control environment [85]. In addition, the environmental and security signals may be added along with tariff signals to get effective performance. In [43, 44], the centralized control on price-based DR program for residential appliance scheduling to minimize the electricity cost and PAR has been discussed. The control suggested in the article [43] provided the cost savings in the prosumers than the traditional end-users, while the article [44] proved the efficiency of the control method in reducing the electricity cost of the consumer including the communication technologies. Residential areas in the city of Bath located in the southwest of the UK have been tested with the DR program [86] and it was concluded that the developed DR program offers both reliability and economical benefits for the utility than the consumers. In [87], a graphical user interface model for two cases such as grid-connected mode and islanded mode has been designed. The centralized control with Particle Swarm Optimization (PSO) technique has been provided the flat load profile for both cases with the utilization of switch agent, central coordinatoragent, local controller-agent, and load agent. In [88], the network has three feeders, the first one serves the residential area, the second one serves the industrial area and the last one serves the commercial area. The overall social welfare of the consumers and the utility have been maximized with the implementation of the centralized control.

2.3.2. Decentralized control

In the decentralized controller, the consumers can change their consumption pattern to afford the demand quoted by the utility. Fig. 3 represents the decentralized control of microgrid architecture. Fig. 3 shows that the microgrid has its decentralized control center. The power generation and power demand are scheduled using this control center where the excess power availability should be communicated to decentralized control centers of other microgrids. In [89], the controller received feedback from the stakeholders of the particular area and the control action has been taken accordingly. The residential and commercial consumer in the area of Vijayanagar at Bangalore city to analyze the demand pattern with the decentralized control. The supply-demand gap has been reduced considerably by utilizing the above-mentioned method. In [90], a decentralized controller was suggested for a residential consumer with installed PV and wind power. The Inclined Block Rate (IBR) pricing scheme for DSM implementation has been suggested. Both the decentralized and centralized controller has been critically compared with various performance parameters in [91] for a residential consumer. The decentralized control was proved to be the optimal controller to minimize the electricity cost as compared with a centralized controller.

2.3.3. Distributed control

Distributed control of the microgrid eliminates the few limitations that present in both centralized and decentralized control techniques. It has a wide range of applications in energy management on the generation and consumer end. Fig. 4 represents the distributed control of microgrid architecture. The distributed control center of the microgrid communicates between the grid and among the other microgrids. The distributed control center decides the power flow to the load based on the power generation of the same microgrid, power availability from other microgrids and power taken from the grid . The advantages of distributed control over the centralized and decentralized control techniques have been compared in [92]. The limitations of centralized control include the high cost of centralized control center installation, minimum reliability and stability. On the other hand, decentralized control is cheaper than centralized control, but it is unstable and unreliable due to the lack of communication links and information sharing. The distributed control for the DSM on domestic consumers has been discussed in [93]. The developed control technique has been resulted in reducing the peak demand, transmission losses and the dependency on the grid.

2.4. Business layer

The tariff would be calculated in the business layer and communicate to the utility. This layer controls the DSM operations and decides the investment to be done on additional plant capacity. The twopart pricing scheme has been introduced which is the combination of centralized control center data with RTP and capacity pricing component. The scheme has been proved to be effective in reducing the peak demand [94]. The twelve tariff components and their impacts on the electricity bill for a community of 100 residential consumers has been analyzed in [95] and proved the best choice of billing system as compared with the volumetric billing system. The net purchase and sale for the billing purposes have been implemented. E+ business model has been framed in [96] based on the two agents such as prosumer and aggregated prosumers manager which would be increased both the electric and thermal energy efficiency. Additionally, it can control the energy level of the whole distribution grid. Section 3 deals with the DSM implementations in the microgrid. Also, the different DR programs such as DLC and indirect load control are discussed in Sections 3.1 and 3.2 respectively.

3. Demand side management

The implementation of the load management technique in the electricity suppliers may reduce the cost during the peak power and eliminate the difficulty of increasing the additional capacity on the generation side [58]. The changes in the conventional load management technique denoted as Utility-DSM (U-DSM) in which the demand pattern can be modified to obtain the flat demand curve in the utility itself without considering the consumers [58]. Modern utility DSM programs have been derived from the national energy conservation policy act [97]. Since 1980, DSM has become very popular.

DSM has been utilized to balance the time-varying demand of consumers and generation capacity of the power systems. It has encouraged consumers to actively participate in energy management by providing the incentives. The consumer end energy management deals with the power flow and information exchange between the utility and the consumer [90]. In [98], Minnkota power co-operative supplies 12 rural electric cooperatives in Minnesota and North Dakota. The annual load factor was improved on these electric co-operatives by proper planning on the demand side for effective load management during the severe winter peaks. The manual switching from electricity to oil for space heaters was adopted during the peak-load hours. The load factor was improved from 48% in 1976 to 63% in 1983 without any consumer discomfort [98]. The modern DSM relies on smart switching



Fig. 2. Centralized control of microgrid.



Fig. 3. Decentralized control of microgrid.

according to the demand and supply relationship. DSM has become popular by means of encouraging the consumer side power generation and modifying the electricity tariffs. A flat power demand curve would be obtained by considering both consumer side power generation and electricity tariff. DSM can be implemented by the consumers in three ways [99]:



Fig. 4. Distributed control of microgrid.

- 1. Energy Conservation: Energy conservation reduces energy consumption at peak hours without any alternative solution to the consumers which causes discomfort on the consumer side. Energy conservation measures in the generation, transmission, primary and secondary distribution were discussed in [100]. The combination of energy efficiency and energy conservation was considered to reduce the power consumption. Energy efficiency measures on residential, commercial, industrial and agricultural sectors in India have been analyzed in [101]. The energy shortage in Gujarat could be removed by saving the annual energy (GWh) using energy efficiency technologies. Also, the peak power saved (MW) on the demand-side implies proportionally lower investment than implementing the new power plants.
- 2. Peak shifting: In peak shifting, part of the energy consumption is shifted from peak hour to off-peak hour such that overall consumption remains the same. Peak shifting is one of the most accepted DSM implementations. Much literature has been discussed peak shifting with the different optimization techniques [79,102]. In [79], a two-stage hierarchical energy management for the residential consumers with semi-scenariobased Rolling Horizon Optimization (RHO) algorithm was suggested to reduce the PAR. Data Analytical Demand Response (DADR) in [102] has also a two-stage DR technique such as the consumer-centric and the utility-centric. User comfort and willingness to participate in DR were given importance. Also, the scheduling of appliances could be done by utilizing Consumer-Centric DADR (CC-DADR) based on the consumers' consumption data from smart homes appliance adjustment factor, and appliances priority index. If the CC-DADR failed to reduce the demand up to the threshold level which has declared by the utility, the scheduling would have updated by Utility-Centric DADR (UC-DADR). The penalty will be paid by the utility for the consumer sacrificing [102]. In [91,103,104], peak shifting was done using the distributed algorithm with 'L1' regularization,

artificial fish swarm intelligence and hybrid approximate dynamic programming and hybrid big bang-big crunch algorithms respectively.

3. Use of On-site generation: DG at the consumer end has advantages such as supports energy management, enhances the reliability of the system and minimize transmission losses. In [105], a grid-connected system consists of a hybrid renewable system with EV has been suggested for the DSM. Five different batteries were compared and a lead-acid battery with a carbon-enhanced electrodes were found to be economical for variable loads [105]. As residential PV penetration is high in Australia, the solar power was utilized for obtaining the flat demand curve in [106]. The capital cost of critical peak availability from gas turbines with the distributed solar in the Australian national electricity market has been compared in [106]. Texas has the largest wind power plant among the other states in the USA which has capable of supplying the power for 71000 MW peak demand [107]. The supply capacity of the wind power plant is greater than its 15 years average value of 64000 MW [107]. In [107], both solar and wind power were preferred as the peak power plants because of zero marginal costs. EVs were also used for supplying the peak demand which operates in charging mode during the off-peak hour (peak shifting) and discharge mode during the peak hour (valley filling) in [108,109]. In [110], it was proved that the real-time decentralized DSM was efficient in reducing the PAR than the centralized scheme of the grid-connected system which consists of the RES such as PV, wind, Energy Storage System (ESS) and EVs. It enabled the consumer to produce a flat load profile and to reduce the electricity cost. The different DSM implementations have been tabulated in Table 2 based on the type of consumers, DG, storage, and application of computing techniques. Section 3.1 discusses the different DLC techniques. Section 3.2 elaborates on the incentive and price-based indirect load control techniques. Section 3.3 explains the soft computing



Fig. 5. DSM — Load configurations.

techniques that have been utilized to implement the DSM in the microgrid. Section 3.4 provides information about the scenario on DSM implementation in India.

3.1. Direct load control

Among the three ways of implementing the DSM, the combination of peak shifting and on-site generation to form the current DSM, while energy conservation is not considered. The DSM is categorized into two techniques such as energy efficiency and DR. China invested in the energy efficiency technique by the replacement of Compact Fluorescent Lamps (CFLs) for lighting and decided to modify the electricity tariff in the year 1990 [90]. The state of California had spent 20 billion USD on new power supply contracts during the power crisis in 1999-2000 [111]. It was only able to increase the electricity supply by 2% which is indicated in the case study presented. Whereas, an investment of 1 billion USD on energy efficiency promotion and modification could achieve a 10% reduction in the electricity demand. But the energy efficiency improvement has not sufficient for load management in many cases. In such cases, DR is widely used as a DSM technique that encourages the consumer to modify the consumption pattern and thus alter the configuration of the load curve. Different load pattern changes are portrayed in Fig. 5. Change in load pattern using DSM implementation is obtained by six different methods such as peak clipping, load shifting, valley filling, load growth, strategic conservation and flexible load curve. From Fig. 5, it is evident that six different ways of modification in the actual load curve could be implemented to reduce the peak load. Peak shaving and strategic conservation are the configurations that reduce the electricity consumption without operating the load. On the other hand, valley filling and strategic load growth are the configurations that discuss the increase in load demand. The flexible load curve has unrestricted load variation in all durations. Peak shifting is the most beneficial load configuration which combines both peak shaving and valley filling by shifting the operation of loads from peak hour to off-peak hour.

1. **Peak clipping:** In peak clipping DR programs, the usage of power during the peak-load hour is decreased [112]. In [113], it has been stated that the microgrid can be actively responded to the main grid, the peak-load of the grid tie-line can be clipped with Peak Time Rebate (PTR) programming. Also, it has

analyzed the peak-shaving indices such as peak-shaving quantity index and peak-shaving ability index [113]. In [114], PTR along with Incentive-based DR program (IBDR) have obtained the peak clipping by optimal scheduling of controllable appliances. BESS has also used for peak demand reduction. The South-European islanded power system with relatively higher load consumption has been considered in [115] which consists of BESS and renewable energy. The available renewable energy power from the South-European island is very low when compared to its peak demand [115]. Artificial Neural Network (ANN) has been developed for peak clipping during the peak demand hour. Participation of customer side BESS providing the peak-load shaving at distribution feeder with the value of 12.47 kV have been analyzed in [116] for both online and offline optimization. Online optimization has been monitored by Supervisory Control And Data Acquisition (SCADA) and the cost minimization have been observed with ToU based tariff.

- 2. Valley filling: In valley filling, customers are encouraged to use their equipment during off-peak hours. In [117], an online algorithm has been developed for scheduling the EV charging during off-peak hour. IEEE 14 bus system has been considered which has five generator buses and nine buses connected to the EV. By varying the charging rate of EVs during the off-peak hour to fill the valley leads to obtain the flat load curve. For IEEE 30 bus system, Grid to Vehicle (G2V) has been scheduled with the water filling algorithm for charging the battery of the EVs [118].
- 3. Load shifting: In load shifting, the electricity usage is shifted from the peak-load hours to off-peak hours. Load shifting is the combination of peak clipping and valley filling. With the increasing number of EVs usage, the load shifting could be done with the effective interaction between EVs and the power grid [27]. EV charging/discharging model has been developed based on the PSO algorithm to achieve the optimal results through the groupbased model in [27]. In [119], the customers have been clustered based on the consumption pattern in which the group with the largest load clipping capability was selected. The load clipping capability was based on the effectiveness of the customer to reduce the supply shortage. Therefore, the load has been shifted to off-peak hour from the peak hour. Peak load shifting in the different applications was tabulated in Table 3.

- 4. Load growth: In load growth, the load-shape changes as an increase in sales by the utility. On the utility side, load growth was linked with the planning for future development and increase in the reserve capacity. For reliable operation of the system, the sequential Monte Carlo simulation method was utilized to forecast the load growth. The optimization model has been tested for the Indian radial distribution system with two scenarios such as DG integration and expansion of reserve capacity. Scenario with DG integration was found to be better than the expansion of reserve capacity [120].
- 5. Strategic conservation: In strategic conservation, total energy consumed over a year is reduced. In [121], illustrates the behavioral changes required for energy conservation. It has also explained the DSM programs in different countries and their objectives. In [115], home automation has been suggested as the energy conservation measure for residential consumers. The demand reductions of 1–5 kW per house during summer and 1–2 kW per house during the winter in the southern region of the USA [115]. Conservation voltage reduction has been suggested for energy savings in [122]. The voltage on the feeders that run from the substations could be reduced while applying the conservation voltage reduction technique in power systems for peak demand shaving and energy consumption reduction. Therefore, it produces considerable energy savings in both utility and consumers.
- 6. Flexible load curve: The electricity usage during different times is redistributed to reduce the PAR in [123]. Interruptible load hopping and constant interruptible load hopping algorithms were applied for hopping and centralized DLC schemes to flatten the load curve leads to reducing the forecasting error. A centralized DLC scheme has been found to produce the flat load profile. In addition to the flat load curve, it has some advantages such as low communication and computation overhead and full consumers privacy [123]. Flexibility in the load shape could be obtained by integration of the grid using dynamic control to schedule the allocation of DG and storage based on the response of the consumer [124]. Different methods of dynamic control are DLC, through independent control agents, using controllable appliances and EMS.

3.2. Indirect load control

DR programs are adopted for encouraging consumers to modify their load configuration. DR programs are classified into two programs such as IBDR programs and price-based DR programs. The detailed classification of DR programs is illustrated in Fig. 6.The major classifications are IBDR and price-based DR. The IBDR is further classified into classical programs and market-based programs. Similarly, pricebased DR is further classified into RTP, ToU and CPP. IBDR is the most commonly used techniques to modify the load pattern.

3.2.1. Incentive based DR programs

IBDR provides incentives to the customers who are willing to reduce/shift their electricity consumption during peak-hours to off-peak hours or a discount in the electricity price is given to these customers for their participation in the program [125]. IBDR or dispatchable programs are classified into two types such as classical-based programs and market-based programs. In classical DR programs, customers receive incentive payments for allowing the utility to control demand by DLC and Interruptible or Curtailable services (I/Cs). In a market-based DR program, consumers receive a discounted electricity rate for reducing the load on request from the utility. Market-based DR programs are classified into four types such as Emergency DR Programs (EDRP), Capacity Market Programs (CMP), Demand Bidding (DB) and ancillary services market programs. In indirect load control, the utility itself reduces the demand of the consumer with an incentive payment to reduce the peak demand [125].

A residential community of 40 smart homes with PV generation and BESS has been taken into account in [43]. It has considered the revenue index and satisfaction cost as the constraints and scheduling have been done with the help of an algorithm based on the Stackelberg equilibrium. It was proved that the outcome of the scheme obtained the flat load profile [43]. Aggregation techniques in this scheme were applied to minimize the loss of welfare of DR resources [91]. On I/Cs, the consumers receive either a discount in electricity bills or bill credits to reduce the load on the occurrence of system contingencies. In EDRP, the utility provides an incentive to the consumers for reducing their loads in case of reliability events. Consumers have an option either to pay the penalty without curtailment of the load or curtailment of the load when the utility notified the customers. The level of the payment is usually notified ahead to the customers from the utility. In CMP, consumers are allowed to commit in providing the load reductions during the system contingencies. If they do not curtail the specified load they have to pay the penalty. The customers may receive a guaranteed payment to curtail the load when it is directed from the utility [91].

In the DB program or the buyback program, consumers bid for a specific load reduction in the day ahead based wholesale electricity market [150]. Consumers have the liberty to choose the value of bidding for the amount of energy reduction. They will be rewarded if the energy saved matches the quoted amount. If it fails to satisfy the requirement, no financial penalty is imposed on the consumer [150]. The last type of IBDR is the ancillary services market programs. It allows the consumer to bid for ancillary services such as storage, operation reserve, reactive power, voltage and frequency control. In [151], the consumer has to bid for acting as reserves by curtailing the load. The consumers were notified by the ISO when the necessity of reserve capacity and they will be paid based on the market price. It disturbs the running load very often. Three cases without DR programs, with RTP and RTP along with IBDR have been compared and it was proved that the IBDR along with RTP provided the flat load profile effectively in [152]. The DSM in the point of retailers have been discussed in [153, 154] and suggested that the IBDR reduced the cost of the customers and maximized the profit of the retailers.

3.2.2. Price based DR programs

Price-based DR programs focus on the minimization of PAR by modifying the tariff structure instead of providing incentives to the consumers. In this type of DR program, the electricity charge is higher than the normal charge during the peak-load hour whereas during the off-peak hour the electricity charge is lower than the normal charge [79]. Price-based DR programs are classified into RTP, ToU, CPP. CPP is further divided into two types such as extreme day pricing and extreme day CPP. Also, the PTR technique is explained at the end of this section.

3.2.3. Real-time pricing

RTP is the most effective and widely used strategy to reduce the peak-load and PAR. Here, the real-time market electricity price is replaced with respect to the real-time generation cost. The price fluctuates with respect to the time would be informed to the consumers by the retailers based on the day-ahead basis or hour-ahead basis price signals. Therefore, they can participate in retail markets with lower risk so that the RTP becomes more efficient. In [79], semi-scenario-based RHO has been used for the optimization of HEMS to reduce the electricity cost and improve energy efficiency. One supplier, one utility, one ESS and 50 residential consumers have been considered to improve the social welfare with the flexible charge rate depends on the time in [125]. Considering the RTP alone could create the temporary peaks during the off-peak hour [125]. In order to avoid the above-mentioned drawback, the combination of RTP and IBR has been suggested in [90]. Consumers are classified into grid energy consumers, smart energy consumers and

Table 2	
---------	--

11

Comparison table of different DSM implementations.

Ref No	Objective	Constraints	Type of consumer	Control	DR program	Sources	Storage	Optimization techniques	Result outcome	Inferences
[43]	Minimization of cost including ESS cost	Residential BESS, controllable appliance operation time and operation cycle, non-interruptable controllable appliance operation and interruptible controllable appliance minimum online/ offline time constraints	Residential, commercial and industrial	Centralized	RTP	PV	Battery (Lead acid, Lithium ion), EV	Optimal scheduling algorithm	Performance of lead acid and lithium ion batteries were compared. Efficiency and depth of discharge of lithium ion batteries are proved to be better than the lead acid batteries	Load scheduling not considered along with resource allocation
[79]	Minimization of residential tariff and improvement of energy efficiency	Power flow and limits of DG busbar, load bus, EV aggregator busbar and voltage ramp limits	Residential	Decentralized	RTP	PV	Battery	Semi scenario based RHO	Cost was comparatively low as compared with RHO method and HEMS & absence of semi scenario based RHO	Heating/cooling loads were not and only limited appliances are considered to minimize the cost
[90]	Reduction in PAR and average waiting time, cost minimization with user comfort	Power balance constraints, SoC of energy storage system	Residential	Decentralized	RTP+ IBR	PV and Wind	Battery	Genetic Wind- driven optimization (GWDO)	The developed optimization has been compared with Genetic Algorithm (GA), Binary PSO (BPSO), Wind Driven Optimization (WDO) and it was found to be more suitable for DSM	Consumer comfort was not considered
[125]	Maximization of welfare including supplier, utility and consumer, selection of optimum charge rate	Power balance constraints, SoC of battery, minimum and maximum limits for power generation and consumption	Residential	Centralized	RTP	-	Battery	Distributed load scheduling algorithm with convex optimization method	The presented optimization technique was found better than optimal RTP algorithm. Flexible charge rate have been suggested to afford intermittency	Load shifting was not considered and centralized control affects the privacy of consumer
[126]	Minimization of electricity cost and peak shifting	Power balance constraints, ramp rate of power at time slot	Residential	centralized and decentralized	RTP	_	-	Optimum stopping rule (OSR) based real-time distributed scheduling algorithm	The presented algorithm was utilized for both centralized and decentralized controllers through the central scheduler	Distributed controllers could be implemented which are more suitable than the centralized and decentralized controllers.
[44]	Maximization of cost savings on trading of energy	Power balance constraints, ramp rate of power at time slot, priority index, interruption time constraints	Residential	Centralized	RTP+ feed-in tariff	PV	-	Load scheduling optimization	Dynamic RTP was compared with the feed-in tariff and proved to be effective in the prosumer based DSM	Load scheduling was not considered. Resource allocation alone could not be efficient in trading of excess power
[127]	Minimization of PAR and dissatisfaction level at user end	Amount of load shaving in each time step, dissatisfaction level, temperature variation	University of Connecticut campus, USA	Centralized and decentralized	DLC	Combined Heat & Power (CHP)	Fuel cell	Alternating direction multipliers based DR algorithm	SCADA with minimum communication requirement have been considered. Decentralized controller was compared with the centralized controller	CHP are not environmental friendly and economical than DGs such as solar PV and wind
[45]	Minimization of electricity bill and PAR	Power balance and appliance operational time constraints	Residential	Decentralized	TOU	PV	-	Cuckoo, GA, BPSO with and without RES	GA-RES was found suitable for reduction in PAR, cuckoo search method with RES have been suggested for cost minimization and PAR reduction	Storage devices were not considered that has the limitation of intermittency and uncertainity of solar PV
[91]	Reduction in cost of energy consumption, daily peak demand and minimization of PAR	Power balance, standby power level, appliance operation time	Residential	centralized and decentralized	RTP+ ToU			Nikaido-Isoda function based relaxation algorithm, Newton method toward centralized coordination	Customers discomfort have been decreased with sparse pattern. A super linear convergence rate has been obtained using l_1 regularization compared with traditional models and algorithms.	On-site generation, battery and EV were not considered
[46]	Reduction in PAR and improvement in energy sharing among neighbors	Power balance, operation time constraints, appliance operation duration, SoC, Charge\discharge efficiency	Residential	Decentralized	ToU	PV	ESS	GA	Group-based and turn-based co-ordination models were compared with the baseline algorithm	Distributed controllers could be more suitable than the decentralized controllers

(continued on next page)

Table 2	(continued).									
Ref No	Objective	Constraints	Type of consumer	Control	DR program	Sources	Storage	Optimization techniques	Result outcome	Inferences
[128]	Minimization of electricity bills and reduce peak demand by controlling the shiftable loads.	Power constraints, operational time constraints, ramp rate limits	Residential	Centralized and decentralized	RTP	-	-	OSR based opportunistic Scheduling Algorithms	Framed algorithm have been implemented in centralized and decentralized schemes and the savings were 35.6% and 35.2% respectively	On-site generation, battery and EV were not considered
[129]	Minimization of electricity bill including the battery system cost	Power balance constraints, charge and discharge rates, SoC, power conversion losses	Residential	Agent based control	ToU	-	Flywheel, Superconducting magnetic energy storage, nickel cadmium,lithium ion, metal air, lead acid, sodium sulfur, Super capacitor, Compressed Air Energy Storage (CAES) and pumped hydroelectric storage	Peak shaving simulator model	Performance of different storage devices were compared. Pumped hydroelectric storage was proved to be the most suitable ESS. Among the batteries, zinc manganese dioxide has provided the maximum profit considering the seasonal variation.	On-site generation and load shifting were not considered
[48]	Minimization of peak load	Charging rate, discharge rate, time constraints, power balance	Residential	-	DG	ΡV	Battery, EV	Improved decision- tree based algorithm	The algorithm was compared with ANN and proved to be most economical. The two different types of EVs considered in this work were Nissan leaf and Tesla. The peak load reduction of 25% and 68.5% have been obtained with Nissan leaf and Tesla respectively.	Lead acid batteries were harmful to environment and load shifting did not include in the appliance scheduling
[130]	Maximization of profit of residential consumer	Power constraints, time constraints, charge and discharge constraints, Response Fatigue Index	Residential	Decentralized	ToU, CPP, Interruptible or Curtailable services (I/Cs), RTP, Emergency DR Programs	PV, Wind	Battery, EV	RWM for scenario generation	Price-based DR schemes were compared with deterministic and stochastic models. As a result, ToU has provided the maximum profit	Distributed controllers were not included which are more suitable for DSM than both centralized and decentralized
[131]	Maximization of profit with minimum investment	Power balance, Power generation and limits	Commercial and industrial	Decentralized	RTP	-	-	Gaseous diffusion model	EMS operates in coordination with the AMI and communication system to reduce the load and to manage the generation and load	On-site generation, battery and EV were not considered
[132]	Maximization of consumer welfare	Power constraints, ramp limit for consumer budget, consumption pattern	Residential	Decentralized	ToU	-	-	Case1: Fixed daily non-base load budget algorithm, Case2:Fixed daily non-base load consumption algorithm	Case 1 was found to the best in improving the social welfare	Distributed controller with RTP was not considered
[47]	Maximization of shared energy storage provider's revenue and minimization of the user-level social cost	Revenue index and satisfaction cost	Residential	Centralized	IBDR	PV	BESS	Algorithm with Stackelberg equilibrium	The algorithm suggested in this work to achieve the flat load profile	Forecast errors, maintenance of secure grid voltage levels fluctuations, and benefits for participating consumers were not considered
[133]	Minimization of peak-load and energy loss	Substation feeder active power limit, operational time constraint, bus voltage limits, feeder current limits, and LTC and SC operating limits	Residential	Centralized	IBDR	PV, Wind	Battery	GA	Three cases were framed based on β and T_{max} values. As a result, $\beta = 1$ and $T_{max} = 12$ hours, provides the optimal solution	EV was not considered, centralized controllers have privacy issues

(continued on next page)

Table 2 (continued).

	(00/11/11/11/01/)									
Ref No	Objective	Constraints	Type of consumer	Control	DR program	Sources	Storage	Optimization techniques	Result outcome	Inferences
[134]	PAR and energy cost minimization	Power generation, voltage magnitude, line flow and power balance constraints	Residential	Decentralized	IBDR		PHEV	Distributed algorithm with Nash equilibrium	The PAR was 2.1 and the energy cost was \$44.77 without the scheduler whereas the PAR reduced to 1.8 (i.e., 17% less) and the energy cost reduced to \$37.90 (i.e., 18% less), when the scheduler is enabled	Load shifting with appliance scheduling was not considered
[114]	Minimization of the impact on customers being affected by load shedding	Strategy enrollment, load clipping and load shifting, daily consumption, system load variation, system reserve, hourly supply shortage relieving, daily total electricity usage, peak and valley periods load variation	Residential, commercial and industrial	Decentralized	ToU	-	_	Mixed Integer Linear Programming (MILP)	Clusters were formed by selecting the participating customers based on their capability of load clipping, shifting, it is proved that customers who efficiently clip its load could manage the supply shortage.	Impact of DGs and storage devices in load shifting were not considered
[135]	Optimization of the peak rate and rebate	Average electricity price, Customer's satisfactory constraint, peak-valley difference constraint, critical price and discount constraint	-	-	CPP	-	EV	PSO	Electricity bill and maximum peak to valley difference have been compared with different number of CPP and the suggested optimization technique was proved to obtain the optimal result even for higher number of CPP	The model is not consumer centric and impact of DGs were not considered
[136]	Minimization of energy consumption and maximization of residential consumer satisfaction	Peak and valley energy price and demand, satisfaction of residents, price elasticity,	Residential	-	ToU, ladder pricing	-	-	GA	With ladder pricing, the energy saving have been improved from 502.378 kWh to 745.999 kWh and the customer satisfaction have been improved from 0.997 to 0.998	On-site generation, battery and EV were not considered
[137]	Reduction in electricity bills, reduction in the power dissipated through the dump load	Load idle constraint, computational interval constraint, pre-emptive constraint, power demand constraint, battery operation constraint, battery boundary constraint, power export constraint	Residential	Decentralized	RTP	PV and small scale wind	PHEV	GA	The annual electricity bill without IRES was \$2337 whereas, the annual electricity bill with the IRES was \$1159	Distributed controllers could be better than the decentralized controllers
[138]	Minimization of the electricity cost and PAR.	Energy consumption constraint, operation timeslots constraint, power balance constraint	Residential, commercial and industrial	Centralized	RTP+ IBR	PV	ESS	GWDO	Reduced electricity cost and PAR was 22.5% and 29.1% in scenario 1, 47.7% and 30% in scenario 2, and 49.2% and 35.4% in scenario 3	Distributed controllers were not considered and impact of EVs in load shifting was not included
[139]	Minimization of electricity cost and PAR, Maximization of user comfort	Energy consumption constraint, lower and upper limit of scheduling horizon, storage limit	Residential	Decentralized	RTP	PV, wind	ESS	WD-GA, WD- Grey Wolf Optimization (GWO) and WBPSO	The electricity cost was reduced by 35.02%, 35.60% and 53.39% on implementing WDGA, WDGWO and WBPSO, respectively and the PAR minimized was 61.30%, 61.43% and 18.89% by considered the same	Distributed controllers could be more suitable for residential DSM than the decentralized controllers

13

Comparison table of peak-load shifting in different application.

Comparison	table of pe	eak-load shi	fting in different aj	oplication.				
Ref No	DG	Storage device	DR technique	Methodology	Environmen- tal/PQ impacts	Application	Outcome	Inference
[140]	PV	Battery	Dynamic pricing	Metaheuristic solver based on rolling time horizon		Home	The performance have been proved to be best as it includes both day-ahead scheduling and real-time control for optimization.	Appliance scheduling was not considered
[141]		PHEV	DLC	Approximate dynamic programming with heuristic priority scheme		Home	A multi-agent based DR have been suggested for utilization of PHEV in peak-load reduction.	Impact of DGs in peak load reduction was not considered
[142]	-	-	ToU	PSO		Home	Non scheduled and scheduled with PSO results were compared. Peak load and cost reduction could be obtained when scheduled with PSO	DGs and storage devices were not considered
[143]	_	-	RTP	Heuristic based evolutionary algorithm	-	Home, commercial, industrial	DSM has been proved to be effective in all type of consumers.	DGs and storage devices were not considered
[144]	PV, DG, Wind	Battery	Dynamic pricing	MIL problem by CPLEX solver	Reduce emission rate	IEEE 33 bus system	The optimal day-ahead scheduling have been analyzed with and without DR implementation. It was insisted that DR techniques can reduce the cost and peak-load.	Impact of forecasting errors and uncertainities were not considered
[145]	PV, Wind	EV	ToU, Capacity Market Programs (CMP), RTP, DLC, IC, EDRP, CPP	Monte Carlo simulation and k-means clustering algorithm		IEEE 39 bus system	DLC and CMP have been proved comparatively effective in peak load reduction along with congestion management	Battery was not considered. EV could not be used as an alternative instead of battery in this case due to its limited availability time.
[146]	PV	Battery	RTP	Uncertainty- aware microgridbased DR		Commercial Building	Kalman filters have been used to reduce the forecasting error. The model has provided the visible reduction in peak-load and cost for seasonal variation.	Short term load scheduling was not considered. Thermal loads has only considered
[147]	PV	Battery	Dynamic pricing	Monte Carlo simulation	Nearly Zero Emission	Nearly zero energy building)	Nearly zero energy building have been upgraded to Km zero energy building in which the cost reduction was achieved without reducing user comfort.	Peak shifting due to appliances other than thermal loads were not considered.
[148]	-	-	DLC	Queuing theory model		Inelastic Home appliances	Dish washer and laundry machines have been scheduled with the electric plugs. It was controlled by the central controller through internet	DGs and storage devices not considered.
[149]	_	EV	RTP	GA	_	Home	Energy virtual network operator have been suggested for peak-load reduction and better utilization of EVs. It has acted as a virtual battery pool.	DGs and battery not considered. EV could not be an alternative for battery as it is only available during non-driving durations

traditional energy consumers based on the generation and consumption of energy. Appliances were classified into traditional and smart appliances. Smart appliances were classified into power elastic, time elastic and essential appliances. The optimal scheduling has been made by Genetic Wind-driven optimization (GWDO) to minimize the electricity cost, PAR and increase the user comfort. As a result, when the demand is increased beyond the pre-specified threshold, a penalty function will be included in the cost equation which avoids peak formation [90]. Day-ahead RTP have been applied with 32 node radial distribution network in Ontario which provides the maximum profit of both supply provider and customer [155].

3.2.4. Time-of-use rates

ToU is the modified form of RTP. In ToU, the rates of electricity price per unit consumption must differ for different time blocks [91]. The rate during peak-hours is higher than the rate during off-peak



Fig. 6. Classification of DR programs.

hours. The simplest ToU rate must contain the two-time blocks namely peak and off-peak. The three-time blocks such as peak, mid-peak and shoulder-peak were framed with sparse patterns in [91]. It focused on reducing the waiting time, number of interruptions and PAR. Consumers were clustered based on the consumption pattern and the centralized control has been adopted with the proper communication. Smart home management systems developed on a multi-agent system platform with four-time blocks such as peak, shoulder, off-peak and low peak hours were considered in [156]. It was implemented with IoT in the Singapore energy system for efficient information exchange and control.

3.2.5. Critical peak pricing

CPP is the DR program in which the customers are imposed with very high prices during the critical peak time [157]. CPP is higher than the peak price but occurs for a very short duration. CPP has four schemes such as fixed-period CPP (CPP-F), variable-period CPP (CPP-V), variable peak pricing and critical peak rebates. The two constraints considered to implement CPP were the number of critical peak times implemented by the energy service provider within a certain period and the minimum interval (e.g. 24 h) between two different critical peak times. When the energy demand significantly increases, the participating customers would be notified about the price increment [157]. In [158], the CPP was implemented in the U.S. New England electricity and the system performance was analyzed. To achieve the performance, the two different schemes adopted based on the real-time basis such as CPP-F scheme has preset critical peak time with days and CPP-V scheme has the duration of the critical peak time for different days. In [159], an appropriate peak pricing window has been designed for Great Britain to implement the CPP. The energy cost savings from CPP have been observed by peak demand reduction. The peak rate was set based on the peak-hours to reduce the peak demand and improve the stability of the system. EV batteries were used to supply the power to the consumer during critical peak time in [135]. As the number of participating EVs increases, the peak-valley difference and the electricity bill have been reduced. The results of the model were verified with the PSO algorithm. CPP with the different base prices was tabulated in Table 4.

3.2.6. Peak time rebate pricing

PTR programs provide monetary rebates to the customers who have reduced the electricity consumption during periods of high-cost electricity (peak times) [163]. The customers who do not reduce the usage of electricity during peak events are simply charged by the normal rate. In [164], the low consumption, off-peak periods and peak periods have been considered as the load profile with the implementation of RTP and PTR. Two cases of PTR have been considered with revenue loss minimization and without loss minimization. It was concluded that the performance of PTR is better than RTP without considering loss minimization whereas RTP is a preferable one with loss minimization [164]. In [165], when the incentive is greater than the retail electricity price, the consumer utilizes more energy during the base-load period which affects the system reliability. It was found that most of the consumers shift/ increase their demand during peak/off-peak hours to maximize their profits. But only the consumers with a high level of uncertainty could reduce their energy consumption. Therefore, it was found that a PTR program may not suitable for a system where the net consumption needs to be reduced [165].

3.3. Load forecasting

Load forecasting is the prediction of future load demand in advance. Load forecasting has been utilized to implement in the day-ahead and hour-ahead DSM programs. Various methods for load forecasting with the implementation of price-based DR programs such as RTP, ToU and CPP have been discussed in [166]. Lacks in the availability of historical data, AMI and communication technologies have been considered as the challenges of load forecasting. The implementation of artificial intelligence techniques in load forecasting has been suggested to handle the historic data [166]. A clustering-based approach has been developed for the analysis of smart meter data to improve load forecast accuracy [167]. The developed forecasting method has been provided the data with minimum errors and reduced the data size.

Load forecasting has been beneficial for load scheduling on both utility [168] and consumer end [169]. In [168], the fast forecast algorithm based on double seasonal exponential smoothing has been suggested to forecast the substation load data in short-term planning. Power outages and fluctuations have not been considered which are the major concern in [168]. In [169], time-series image encoding techniques and convolutional neural network have been utilized to forecast RTP and power demand. The effectiveness of the forecasting has been proved based on the different parameters such as mean absolute error, root mean square error and mean absolute percentage error [169].

The application of Deep Reinforcement Learning (DRL) has minimized the forecasting error by utilizing Q-network [170]. The peak shifting has been achieved based on the demand and generation forecast obtained from the DRL [171]. The impacts of EV in peak shifting have been implemented based on the load forecasted from the RL technique [172]. But the major limitation with the load forecasting is the uncertainties in load forecasting and forecasting errors were not considered while discussing the impact of EV.

Critical Peak Pricing with different base price

Ref no	Base price	Objective	Optimization technique	Description	Inference
[157]	RTP	Maximization of incentive and profit of energy service provider	Backward dynamic programming	Swing option methodology have been used in the utility to decide the time of optimal critical peak points.	The developed peak pricing has not considered the profit of consumer
[158]	TOU	Reducing the customer electricity bills and electricity purchase cost of utility	Monte Carlo simulation method	In CPP-F scheme, the duration of critical peak period, critical peak rate and maximum number of critical peak hours per year were predetermined	In CPP-F scheme was not flexible where only critical days can be selected
[159]	TOU	Reducing the demand during critical peak time and the electricity cost	Statistical model	The peak/off-peak rate have been calculated based on the market volume and market price data. It follows the CPP-F	Transmission cost, operation cost and distribution cost were not considered
[160]	TOU	Maximization of social welfare	Quadratic programming	CPP of 6.3504 Baht/kWh was proved to be optimal for Thailand Power system	Forecasting error and uncertainity were not considered for tariff prediction
[161]	TOU	Minimization of electricity cost and optimize the load curve	Non-sorting GA -II	In commercial building of Taijan, 15% expense was reduced, 20% peak-load was curtailed, 35% peak-valley gap was filled with the developed method. The method has also provided expense saving of 8% and 30% of peak-load shed when executed the strategy in peak days.	The developed CPP was not suitable for non-peak days
[162]	RTP	Minimization of generation cost and emission rate	Modified <i>e</i> -constraint method	For a 10 generating unit test system with centralized control at ISO the efficiency, of the CPP load control have been proved effectively.	Impact of uncertain sources such as PV and wind were not considered while setting the tariff

3.4. Energy management

Energy management in the consumer end has been achieved by DSM implementation. ToU-based DR has been implemented in a microgrid with combined heat and power (CHP) plants [173]. Robust optimization has been utilized for scheduling the CHP plants to minimize the electricity cost. The results have been proved that the efficiency of the developed framework in minimizing the electricity cost for different cases in [173]. The CHP plants have supported the thermal demand of the consumers which indirectly reduces the electrical demand. The intra-day DR has been implemented for Hybrid Power Plant (HPP) with storage devices in [174]. The HPP consists of wind, PV, compressed air energy storage (CAES), BESS, and thermal units. Hence, the electrical demand due to the heating loads has been supported by thermal units and the remaining electrical demand has been supported by the PV, wind and storage devices. The scheduling of HPP has been framed into a multi-objective optimization to maximize the profit of HPP and conditional value at risk. The risk-based DR framework has been suggested to schedule the reserve capacity of the Virtual Power Plants (VPP) in [175]. The VPP consists of DGs, BESS and flexible loads. The developed bidding-based electricity market model has been achieved the maximization of VPP profit in [175]. The reliability of VPP has been improved by handling the uncertainty of the DGs in it. The risk assessment-based DR has been implemented in the microgrid with CHP, CAES, RES and thermal energy storage [176]. The multi-objective optimization has been modeled to minimize the electricity cost and carbon emission rate of microgrids.

3.4.1. Residential consumers

Different methods of implementing the DSM on HEMS have been discussed in [177]. It has been insisted that space heating should

be considered for peak load reduction. Scheduling of thermal loads alone has been considered in [177]. RTP-based DR program has been considered as the most suitable. The RTP-based DR program has been considered to minimize the electricity cost and peak load of residential consumers in [45,126]. HEMS with RTP-based DSM has been utilized to minimize electricity cost and improve energy efficiency [178].

3.4.2. Industrial consumers

DSM could be implemented in industries where it contains timeinsensitive processes. In [179], peak load has been decreased by 19% and the electricity cost has been reduced by 4% due to the shifting of spinning process in the cotton mill from 08:00 to 09:00 h. In [180], the air separation plant has been implemented with a ToU-based tariff. The peak load of the plant has been reduced which in turn reduced the electricity cost of the plant [180]. In [181], the food and plastic sector have been considered as the smart consumers where the electricity cost has been reduced by scheduling the heating, cooling and air conditioning loads.

3.4.3. Commercial consumers

Energy management in the commercial sector has been limited to thermal storage scheduling in [182]. In [183], peak load reduction and CO_2 emission reduction have been obtained by scheduling the thermal storage and thermal loads. Capital cost has been insisted as a barrier to implement the DSM in commercial buildings [184]. In [185], the energy trading between two prosumers has been achieved a cost savings of 4.9%. The detailed literature survey based on energy management with different kinds of consumers has been critically discussed in Table 2.



Fig. 7. Classification of optimization techniques.

3.5. Soft computing techniques for DSM

The optimal solution of DSM can be achieved effectively by implementing the soft computing techniques. The mathematical models for selecting the objectives based on the appropriate application have been discussed [186]. Also, the different algorithms with mathematical and behavioral modeling have been examined for various applications in [186]. The different types of optimization techniques used for solving the objective function as shown in Fig. 7. Optimization problems are classified into mathematical modeling and application of optimization algorithm. The mathematical modeling is categorized based on the type of objective function whereas optimization algorithms are classified based on the inspiration from the environment to explore and exploit. The objective of the DSM mainly focuses on the minimization of PAR, minimization of cost, maximization of social-welfare, minimization of emissions and minimization of losses [187]. In [187], optimization-based residential energy management has been compared with in-HEMS. The different DR programs and the application of artificial intelligence, neural network and ML in the DR techniques have been reviewed in [188]. The DSM was implemented with artificial intelligence-based elitist non-dominated sorting GA II resulted in a cost-saving of 51.4% when including the DG and battery in [189].

Based on the application, constraints and complexity of the microgrid, a suitable mathematical model can be selected. The major classification of the model includes linear and non-linear programming [190]. The minimization of electricity bills and consumer discomfort have been focussed in [190]. Linear programming can be further divided into Integer Linear Programming (ILP) and MILP. ILP for efficient energy management of the integrated operation of microgrid networks [190, 191]. In [192], a two-phase planning model was framed with MILP for optimal operation of DG. The DSM for a grid-connected residential consumer having the own PV energy generation has been presented in [193]. The scheduling for cost minimization has been done with the help of dynamic programming.

Heuristic algorithms are widely employed where the approximate solutions required whereas the finding of exact solutions requires more time and computationally expensive [194]. Meta-heuristics have been utilized to overcome the drawback of the heuristic approach like to attain the difficulty of local optima. The most commonly used metaheuristic algorithm for DSM is the Genetic Algorithm (GA) that solves the problem through the concept of chromosomes. GA has been utilized to obtain optimal economic operation of the system in two stages. Gujranwala electric power company located in the hub of an industrial and commercial zone that suffers from a huge supply–demand gap have been considered in [195] with neural network for scheduling. The objectives were framed to reduce the supply–demand gap, minimize the black-outs, increasing efficiency, increasing revenue, enhanced reliability and consumer trust/satisfaction without huge investment and wastage of time [195]. Model predictive control implementation on building owners, energy managers or energy-saving companies has been increased both energy savings and occupant's comfort [196]. Also, the developed method reduces the user's complaints and energy bills. In [142], a test system with six household appliances with the PSO technique has been utilized for demand shifting from peak-hour to off-peak hour. The recently developed nature-inspired optimization algorithms for DSM implementation have been critically discussed in Table 5.

A two-stage DR scheduling was developed in [207] for increasing the cost savings and improving the system performance. The developed real-time framework was proved efficient for improving the performance where customers can participate in different DR schemes. It has also predicted and secured the system from dishonest entities using ML. The ML methods such as Q-learning, DL, RL and DRL have been widely suggested for DSM implementation in microgrid. Table 6 critically discussed the application of different ML techniques in DSM implementation.

3.6. DSM in India

In India, the DSM initiative was taken by implementing the energy efficiency measures in residential consumers under control of the Bureau of Energy Efficiency (BEE). BEE has framed the plans and policies to implement the utility-driven DSM. In [215], the DSM programs in India have been critically reviewed and compared with the DSM programs in UK and USA. Also, the barriers to the implementation of DSM have been discussed and the necessity for institutional frameworks in DSM implementation was insisted in [215]. The DSM is further extended by implementing the Time of Day (ToD) tariffs for industrial consumers by State Electricity Regulatory Commission (SERC) [216]. Each state in India follows its own tariff. Table 7 provides the ToD tariff implemented in different states of India. In 2016, the Government of India amended the tariff policy to promote DSM by providing the initiatives to the consumers. In 2017, the SERCs has initiated to implement/install smart meters for consumers having consumption of greater than 500 kWh and consumers with greater than 200 kWh consumption at the end of 2019 [216]. Also, the Government of India encourages the SERCs to consider the renewable energy plants installation at the consumer end.

India's first large-scale DSM program was initiated in southern and western Delhi by Bombay suburban Electric supply Rajdhani power limited and oracle utilities (Opower) with the support of United States trade and development agency. For this program, 2.6 lakh consumers were selected from 10 different districts where the overloaded occur in

Def No.	algorithms used for the impleme	Objective	DB	Description	Terformer
Ref No	algorithm used	Objective	DR program	Description	Inference
[197]	GA	Minimize the electricity cost and peak load	RTP	Appliances scheduled using GA to minimize the electricity cost to 29% and peak load reduction of 36.2% as compared with loop search algorithm	Thermostatically controllable loads were not included in the scheduling
[198]	Real coded GA	Minimize the peak load, maximize the profit of power producer	RTP	Power demand has reduced and profit of power producer & load factor has increased which tends to flatten the load profile	Consumer preference and response were not considered
[199]	Grasshopper optimization algorithm	Minimize the generating and operational costs of microgrid	Dynamic pricing	The profit of the microgrid has been improved by scheduling the resources. The algorithm has been proved to be comparatively effective than whale optimization and cuttlefish algorithm	Load shifting was not considered which may further improve the cost savings
[200]	Bat genetic algorithm	Minimize the electricity cost, PAR and appliance delay time	ToU	The Bat GA has been proved to be efficient than GA and Bat algorithm	Minimization of cost but increased the delay time. Suggested DSM model was not achieved both minimization of cost and delay time at the same time
[201]	Hybrid genetic and pigeon inspired optimization algorithm	Minimize the peak load, PAR and electricity cost	RTP	The hybrid genetic and pigeon inspired algorithm has achieved the most optimal cost for different operational time intervals than GA and pigeon inspired algorithm	Minimization of cost but increased the waiting time which in turn affects the user comfort
[202]	Hybrid elephant and firefly optimization	Minimize the electricity cost and maximize the user comfort	RTP, ToU and CPP	Developed hybrid algorithm has achieved the optimal electricity cost and waiting time than the elephant herding algorithm and firefly optimization algorithm. In particular, RTP was proved to be best in minimizing the cost whereas CPP was proved to minimize the waiting time	Limited appliances were considered and consumer comfort was sacrificed
[203]	Bacterial foraging and Flower pollination algorithm	Minimize the electricity cost and PAR	СРР	This work has achieved with minimum waiting time by utilizing the hybrid bacterial foraging and flower pollination algorithm	Limited number of appliances were considered as schedulable while most of the appliances were categorized as non-interruptible and unschedulable
[204]	Adaptive moth flame optimization	Minimize the peak load and operational cost	DLC	The developed DSM program has achieved the cost minimization and peak load reduction in residential, industrial and commercial consumers where the residential consumers has benefited more than other two consumers	Moth flame algorithm has not achieved the cost minimization as compared with PSO algorithm
[205]	Hybrid genetic ant colony optimization algorithm	Minimize electricity cost, carbon emission, and peak load	RTP	The efficacy of the developed DSM model has been proved by comparing the results with GA, WDO, teaching learning based algorithm, teaching learning genetic algorithm.	Appliances were very limited and centralized controller could be replaced with distributed controller
[206]	Modified grey wolf optimization algorithm	Minimize the electricity cost	ToU	Uncertainty on cooling, heating, wind speed, solar irradiation and the energy carriers prices has been included	Appliances other than thermal loads were not considered

D. Kanakadhurga and N. Prabaharan

Renewable and Sustainable Energy Reviews 156 (2022) 111915

Table 6

Application of machine learning in DSM implementation.

Application	of machine leaf	ning in DSW inip	nementation.					
Ref No	Data science methods	Application	Consumer	Objectives	Optimization model	Description	Outcome	Inference
[208]	Supervised ML	Load shifting	Residential	Minimize the electricity cost	Predictive optimization model	Thermal energy storage has been scheduled to minimize the electrical energy consumption. The efficacy of developed model has been proved with mean absolute error and root mean square error	Achieved 40% of cost minimization and 39% of carbon emission reduction	Appliances other heating/cooling loads not considered
[209]	Resilient ML	Energy con- servation with improved security to prevent dishonest entities	Residential	Minimize the electricity consump- tion	Predictive model with Naïve Bayes algorithms	The developed IoT has enabled ML provided the secure communication with resilient agent	Predicted secure communication correctly — 96%, Predicted insecure communication correctly — 93%, Wrongly predicted as secure communication — 4% and wrongly predicted as insecure — 7%	Limited appliances has considered (only four)
[210]	Supervised ML	Load forecasting	High-rise office building	Minimize the peak load	Data-driven learning algorithm	The accuracy of demand prediction performance has achieved with minimum historical data	Peak load has reduced by 21.9%, normalized mean bias error >10% and root mean square error >30%	Thermal loads of the building were alone considered. Appliances load pattern was not considered
[211]	RL	Peak shifting	Residential	Minimize the peak load and electricity cost	Q-learning with Markov decisive process	The EV SoC has been maintained along with the load shifting	Total electricity cost has reduced by 20% and peak load has reduced by 24%	Limited appliances were considered as controllable loads and data dimensionality is a major drawback of RL
[172]	RL	Load forecasting	Residential	Minimize the peak demand	W-learning	RL has been utilized to forecast the load pattern and EV usage to implement peak shifting	EV peak charging has reduced by 33%	Forecasting errors and uncertainities were not considered
[212]	RL with time delay	Energy con- servation	Isolated microgrid	Minimize the load consump- tion	ANN	RL has been utilized to minimize the power consumed by electric water heaters in microgrid	Average energy consumption by electric water heaters has reduced by 8.91%	Appliances other than water heater were not considered
[213]	DL	Load and generation forecasting	Pennsylvania- New Jersey- Maryland microgrid	Minimize the forecasting errors	Recurrent neural network	Solar PV & wind power generation and load pattern has been forecasted with stacked long short-term memory to implement DSM	Mean absolute error, root mean square error and mean absolute percentage error are obtained as 0.021, 0.36 and 0.95 respectively	Load shifting was not considered. Performance has to be improved with implementation of energy trading
[170]	DRL	Load and generation forecasting	Distribution network	Minimize the forecasting errors	Deep neural network	Deep Q-network has been utilized to improve the privacy	Error distributions for 2%, 4%, 6%, 8% & 10% are obtained as 0, 0, 0.61%, 0.98%, & 1.28% respectively.	Impact of interaction between distribution companies and DR aggregators on privacy was not analyzed

(continued on next page)

D. Kanakadhurga and N. Prabaharan

Table 6 (continued).
-----------	-------------

Ref No	Data science methods	Application	Consumer	Objectives	Optimization model	Description	Outcome	Inference
[171]	DRL	Peak shifting	Residential	Minimize the peak load and maximize the cost savings	ANN	DRL algorithm has been utilized to reduce energy consumption by optimizing the usage of PV energy production	Cost savings has achieved by 16% and 10.2% of peak load shifting has done	Thermal loads were alone considered. Privacy issues were not addressed
[214]	DRL	Peak shifting	Residential	Minimize the peak load	Deep deterministic policy gradient	DRL with disappoint coefficient and penalty coefficient has been considered to improve the performance in load shifting.	Disappoint coefficient was obtained as 0.62 which is the loss due to failure of task. Penalty coefficient has zero which indicates no penalty for DSM violation	Impact of security and privacy issues were not addressed

Delhi [217]. They were randomly divided into two groups based on the information exchange. Two lakh consumers receiving the home energy report from the first group whereas the second group of sixty thousand consumers did not receive the home energy report [217]. Based on the communication mode, the consumers with home energy reports were further classified into three groups such as a paper-based report, web/communication and email-based communication. The program encourages the SERCs of remaining states in India to implement the same kind of programs in their own states. Section 4 includes the contribution of ESS in the DSM implementation. Section 4.1 explains the part of EV as a storage device in DSM. Section 4.2 discusses the necessity of BESS in the implementation of DSM in the microgrid.

4. Peak shifting based on Electric Vehicles and Battery energy storage system

4.1. Electric Vehicles

International energy agency in [218] stated that "Electric vehicles can save more energy than they use". It has indicated the various benefits of EV in both economical and environmental aspects [218]. The different types of fuel cells and their features, storage elements (batteries or fuel cells), configurations, energy management and scheduling have been critically reviewed analyzed in [219]. The Fuel cell-based EV has been suggested as a future EV that controls the emission rate and considerably long driving range [219]. The components and controls associated with the EV have been discussed in [220]. It has also stated that the performance of EV could be better by focusing on the streamlined maintenance management [220]. By optimizing the charging cycles of an electric car using DSM the peak shifting could be obtained. The EV with its charging flexibility has been used for DSM in [221]. Sliding mode control has been used to balance the supplydemand gap despite the uncertainties. The limitations of the EVs have been addressed in [222] and provided the solution to slow down the battery degradation using the efficient charging method. It has also discussed the impact of EVs in high renewable energy penetration regions on supporting the grid.

EVs have been effectively used with RES to achieve various factors such as financial savings, reduction in the demand of thermal generation plant, peak-load reduction and increasing the demand on the renewable sources in [223]. From the results, it was evident that significant gains can be achieved without any further technological advents [223]. An optimization-based model has been presented in [224] to perform the load shifting and congestion management in smart grids during peak hours. Agents called EV aggregators have been suggested to control the charging/discharging of EVs based on the analysis of the Vehicle to Grid (V2G) capabilities [224]. The above conditions have been implemented in a test system based on the IEEE 37-bus distribution grid and the results were proved the effectiveness of the approach on flattening the load curve [224]. In [225], the power interaction between EV and the grid has been presented. A framework for V2G has been suggested to support the bidirectional power flow. It has also discussed the role of plugged-in EV aggregators, different modes of interconnection and the intelligent communication link.

Charging/discharging strategy to control the scheduling of driving pattern has been presented in [226]. EVs at the same bus of a radial distribution system were allowed to charge or discharge at the workplace or home during off-peak hours and peak hours respectively. A multiobjective multiverse optimization algorithm has been applied to decide the number of EVs and the bus at which they must be connected. It was observed that the impact of charging/discharging of EVs was reduced, the cost of electricity was minimum for the users and PAR reduced on the utility side [226]. G2V and vehicle to building have utilized a battery of the EV to supply the required local load in [227]. It has considered two different EVs: Chevy Volt a plugged-in EV model and Nissan Leaf a battery EV model. The performance has been analyzed for six different drive cycles. The load shifting and peak load shaving reduced the peak-load and total energy consumption which reduce the electricity purchase cost for the customer and vehicle owner [227]. A battery swapping station has been presented as an alternative solution for the charging station in [228]. It has been used to manage both high penetration of variable RES and real-time EV charging using the flexible energy storage with DR [228]. Battery swapping station has been used for the smart community microgrid and proved to be more economical.

4.2. Battery energy storage system

In electricity storage technologies, batteries have been widely used at small and medium scales due to their relatively high energy density, lack of geographic constraints, low noise levels, and low maintenance requirements [229]. It is used for obtaining the DSM of peak shifting. Various literature has presented that batteries as effective DR technology. Considering the secondary substation level of the UK electricity distribution, the storage installed at the scale of 1 kWh per house [229] provides the maximum peak demand reduction at a scale of 0.5 kW. Self-consumption of the residential PV, storage units has been presented to compensate for the peak demand in [230]. It is suggested to utilize a community energy storage unit instead of separate storage units in houses [230]. The increment in profit by changing from individual to shared electricity meter in the community obtained was 23% [230].

ToD tariff in different states of India.

States	Peak		Off-peak		
	Duration	Tariff	Duration	Tariff	
Tamilnadu	6 h to 9 h and 18 h to 21 h	120% of normal charge	22 h to 5 h	95% of normal charge	
Andhra pradesh	10 h to 14 h	(Normal charge + 0.75) Rs /kwh	22 h to 6 h	(Normal charge - 1)	
rinding prodesir	18 h to 22 h	(Normal charge + 1) Rs /kwh		in Rs /kwh	
Delhi	14 h to 17 h and 22 h to 1 h	120% of normal charge	4 h to 10 h	80% from normal charge	
Karnataka	6 h to 10 h and 18 h to 22 h	(Normal charge + 1) Rs /kwh	22 h to 6 h	(Normal charge - 1) in Rs /kwh	
Gujarat	12 h to 17 h and 18.30 h to 21.30 hrs (April to october)	(Normal charge + 0.8) Rs /kwh (up to 300 kW (Normal charge+1) Rs/ kwh	22 h to 6 h	(Normal charge –0.3) in Rs /kwh	
	8 h to 12 h and 18 h to 22 hrs (November to march)	(above 300 kW)			
Jharkhand	6 h to 10 h and 18 h to 22 h	120% of normal charge	22 h to 6 h	85% from normal charge	
Haryana	5.30 h to 8 h and 17.30 h to 22 h	(Normal charge+2) Rs./kwh (up to 30% average consumption) (Normal charge+4) Rs./kwh (above 30% of average daily load)	22 h to 5.30 h	75% of normal charge for 20% of average daily consumption	
Maharashtra	9 h to 12 h	(Normal charge+0.8) Rs./kwh (Maharashtra State Electricity Distribution Company Limited (MSEDL))	22 h to 6 h	(Normal charge-1) in Rs./kwh (MSEDL)	
		(Normal charge+0.5) Rs./kwh (Other DIStribution COMpanies (DISCOM))			
	18 h to 22 h	(Normal charge+1.1) Rs./kwh (MSEDL)		(Normal charge+0.75) in Rs./kwh	
		(Normal charge+1) Rs./kwh (Other DISCOM)		(Other DISCOM)	

For different regions of Canada, the battery sizing have been done to obtain the peak demand reduction in [231]. Peak demand reductions between 42% and 49% have been achieved for all Canadian regions except Quebec [231]. These reductions have been achieved when a BESS was installed with an energy capacity of 5 kWh in the Atlantic, Prairies, and BC regions, and an 8 kWh capacity in Ontario [231]. But the peak reduction in Quebec was only 28% because the house in Quebec requires a BESS capacity of 22 kWh [231]. Hence the peak demand reduction in this region is ineffective with BESS .

Grey Wolf Optimization (GWO) has been used for optimally locating the energy storage in [232]. It has considered a 34 bus test system with wind power generation and battery [232]. As a result of implementing the GWO, the minimum battery size required has been calculated and the loss minimization has been improved when compared to the GA by peak shaving at the optimal locations [232]. Optimal charging/discharging regime significantly reduced the electricity bill of the home installed with BESS and PV system [233]. The annual electricity bill has been decreased by 58.65% [233]. Linear programming based optimal sizing of the BESS have been modeled based on the local demand and billing scheme in [234]. It was described that the peak shaving technique could reduce the payback period, return of investment and battery aging of the system in industrial customers [234].

The concept of mobile BESS has been presented in [235] and its performance has been compared with static BESS. The mobile BESS has been improved the economic and reliability efficiency of the system. During the peak season, a mobile BESS can be connected to a peak demand area to balance the demand and renewable energy whereas during the off-peak season the mobile BESS could be moved to another area where the peak-load exist [235]. The scheduling has been made with ANN and proved that the cost savings is maximum. The different ESS such as batteries, fuel cells, electrochemical capacitors or supercapacitors including their advantages, disadvantages and limitations have been discussed in [236]. As a consequence, the different types of EVs could be combined to achieve high efficiency and sustainability [236]. This approach could be effective and more appropriate to reduce energy losses, costs, environmental impacts and health concerns. Section 5 elaborates the advancement in communication, metering and data handling for DSM implementation in the microgrid.

5. Application of Advanced Metering Infrastructure (AMI) and communication technologies in DSM

5.1. Communication and data handling

The advanced technologies in communication, data handling and metering reduces the complexity of DSM implementation in microgrids. IoT and google firebase cloud had developed for the implementation of DSM in residential consumers [237]. The load pattern data had uploaded to the web portal in terms of voltage, current, power and energy that could be downloaded and used for DSM implementation. Open access of the load pattern data is insecure and often possess intrusions. In order to make the DSM more secure and flexible, the ML for the IoT enabled home area network has been suggested in [209]. The real-time problems on data handling have been critically reviewed in [238]. Fog and edge computing has been suggested as the energy efficient methods with the minimum number of data transmissions in [238]. The ToU tariff-based DSM has been implemented using IoT technology with Wi-Fi network connectivity [239]. IBR tariff has been included along with ToU for peak shifting to minimize the electricity cost in both summer and non-summer seasons. Cloud storage-based edge analytics has been utilized for IoT-enabled smart homes to minimize the electricity cost [240].

The energy sector handles a bulk quantity of data such as consumer energy consumption, weather and data from the geographic information center [241]. For DSM implementation in the microgrid, the data storage and handling has been done by big data analysis in [241]. Big data analysis implemented in a building automation system of Hong kong achieved balanced load scheduling despite the huge load dynamics in [242]. It has been utilized an unsupervised data mining technique to perform the operation by automatic knowledge recovery. In [243], big data has been implemented for commercial and residential energy building in the United States. The data collected from the consumers have been recorded and utilized for DSM implementation.

Blockchain is the most emerging technology for data handling in DSM implementation [244]. It has been proved that the real-time application of blockchain in DSM resulted in energy savings and load scheduling without manual operation. The blockchain has been utilized for peer-to-peer energy trading among the prosumers in [245]. The developed blockchain-based trading could encourage passive prosumers to involve in energy trading. Permissioned blockchain-based residential energy trading system has been implemented for a community with 8 homes in a Canadian microgrid [246]. Based on the test system result, it was evident that the peak demand has been reduced and the energy trading has provided the minimum electricity cost. In [247], the contribution of blockchain and artificial intelligence in different energy management processes has been analyzed. Secure data analytics, privacy, smart contracts, development cost and lack of standards & professionals have been given as a future scope for the researchers who works in the blockchain-based DSM implementation [247]. Ethereumbased blockchain has been utilized to implement the DR programs in [248]. Also, a distributed ledger has been recommended as the data storage for secure energy trading in [248]. The contribution of bigdata and blockchain in DSM implementation has been elaborately discussed in Table 8.

5.2. Advanced metering infrastructure

One of the major challenges in DSM implementation is the lack of metering infrastructure [279]. The necessity of smart meters and their privacy issues have been explained in detail [280]. The key issues were lack of data management facilities, lack of prioritizing the privacy for billing & other value-added services and lack of consideration of privacy in economic models. In [281], the lack of privacy-preserving schemes has been analyzed for secure communication between the consumer and utility. Based on the survey results, privacy has to be given importance in both data management and trading of electricity. Also, the cost of smart meters is a major concern. The implementation of smart meters in the consumer premises enhances the implementation of DSM to monitor the key performance indicators such as consumption pattern, savings target and power consumption. In [282], AMI protocol has been developed for improving the performance in terms of computation and communication. The features of AMI are monitoring and recording of energy usage that has been used for DSM implementation in [283].

In [284], the adaptive compression mechanism has been implemented in the smart meter to minimize the communicated data with the grid. It also facilitates the consumers to modify the load pattern using the load shifting technique. The adaptive compressing mechanism in the IoT-based DSM minimizes the burden on grid by compressing the data flow [284]. Smart Compact Energy Meter (SCEM) has been developed to implement DSM in commercial building energy management [285]. The developed SCEM has synchronized with IoT to communicate the demand & tariff, to monitor and control voltage, current, power factor, frequency & power. Also, SCEM notifies the power quality issues to the consumer by text messages. In [286], AMI has been utilized to implement price-based DSM. AMI has connected with the DISCOM & consumers to provide the necessary data to the energy management controller. The energy management controller has been implemented with hybrid bacterial foraging and PSO algorithm to minimize the electricity cost, PAR, CO₂ emissions and user discomfort [286]. The prepaid meters have been suggested to implement the DSM by utilizing the IoT in [287]. Advancements in smart meters and their application in DSM implementation have been elaborately discussed in Table 9. Section 6 deals with the impact of DSM implementation in the microgrid due to PQ issues.

6. PQ issues related to DSM

The major PQ issues on the DSM are voltage deviation, harmonic distortion and phase unbalance [305]. A test microgrid with American voltage levels has been considered and simulated to handle the PO indices (voltage deviation, harmonic distortion, phase unbalance) along with the peak-load electricity cost reduction. The PQ improvement indices have been included in the optimization-based EMS algorithm as the constraints. This algorithm has been included the regulation loop between MILP optimization and harmonic load flows. The active and reactive powers have been included to show the changes in the magnitude of PQ-related parameters in MILP optimization [305]. The IEEE 1547-2003 standard recommends that a microgrid should not deviate voltage variations greater than $\pm 5\%$ around the nominal value [306]. The IEEE 1547.4.11 recommends that at least one DER should be responsible for regulating voltage and frequency in islanded mode while connecting in coordination with other loads and DERs. Also, the voltage unbalance factor is the measure of phase unbalance which must be lower than 3% at every node. Different standards on the %Total Harmonic Distortion(THD) of current and voltage have been addressed in [307]. In, the PQ parameters such as %THD of voltage and current, long-time flicker severity and power frequency have been discussed. A multilayer neural network as a tool for forecasting has been used in [308]. The PQ issues in DSM is illustrated in Fig. 8. From Fig. 8, it is evident that the voltage sag and swell are caused by the intermittent nature of DGs and the switching of loads. Switching on bulk loads at the same duration causes voltage sag whereas switching off bulk loads causes voltage swell. Similarly, bulk switching ON/OFF of similar kinds of loads at the same duration causes the PQ issues such as harmonics and power factor.

The utilization of CFL for improving energy efficiency leads to an increase in PQ problems [309]. The capacitive nature of CFLs has reduced the overall reactive power demand of the load and increased the system voltage. The implementation of plenty of CFLs in the system may reduce the %THD value which is closer to the 5% limit as per the IEEE 1459-2010 standard. Therefore, the performance is found to be better when compared to incandescent lamps. The true power factor of the aggregated load can reduce slightly which leads to reduce the power transfer. The PQ issues like voltage unbalance and harmonics have been created in the plug-in EVs where the EVs were not equally distributed to the phases during the charging period [310]. Therefore, the limit of negative voltage unbalance may exceed the rated value. It was suggested that only a three-phase charging can be used to avoid the above-mentioned drawback [310]. Also, the fifth-order harmonic voltage was slightly reduced by the EVs, but the penetration of EVs dominates the third-order harmonic emission [310]. The different PQ issues in microgrids and their causes have been examined in [311]. Table 10 represents the DSM impact on PQ. The PQ issues in the microgrid are mainly due to the switching of loads and DER [311]. The reconnection of loads creates both voltage variation (sag) and

D.
Kanakadhurga
and
N.
Prabahar

Contribution of big data and blockchain in DSM implementation.

Ref no	Objective	Consumers	Technology used	Contribution of technology	Data	Privacy issues	Inference
[249]	Improve the thermal comfort of consumer, minimize the energy consumption, PAR and carbon dioxide	Residential	Blockchain	Probability of the next hour	Load pattern, initial values and boundary conditions	Privacy settings can be modified by individual consumers through permissioned private blockchain	Limited number of appliances were considered and DGs were not considered
[250]	Flatten the load curve	EV charging network	Blockchain	Peer-to-peer energy trading	Availability of sustainable energy, load pattern, incentives for load shifting	Permissioned blockchain enhances the privacy of the network	Modeling based on uncertainities were not considered
[251]	Eliminate the third party-based system operator	-	Ethereum based blockchain	Energytrading	DG and load power data, storage devices availability	Due to the elimination of third party interference, the privacy is enhanced	Modeling has not been done for consumer specific
[252]	Minimize electricity cost, emission cost and PAR	Smart microgrid	IoT	ToU based DR	Availability of DGs, battery & EV and load pattern	-	Privacy issues with data sharing and handling has not been addressed
[253]	Maximize the welfare of EVs	Plug-in hybrid EV	Consortium blockchain	Double auction mechanism based trading	Battery data, cost for accessing charging lots, Welfare due to V2G and electricity consumption pattern	Consortium blockchain enhances the privacy and security of data	Execution time is large
[254]	Minimize PAR, flatten the load profile and maximize the welfare of consumer & system	Residential, industrial and commercial	Blockchain	Decentralized DSM for peer-to-peer energy trading	Solar, wind, battery & EV powers, Demand profile and tariff details	Privacy against the end users data accessibility	Limited privacy due to the sharing of data only with central control center
[255]	Improve bidding efficiency and social welfare	Buyers and sellers of energy market	Decentralized blockchain	Maintaining the energy contract based on the request from buyers and sellers	Historical data of load pattern and power generation	Using historical data for real-time transactions enhance privacy of the network	Existence of forecasting errors
[256]	Prevent electricity theft and mitigate false data injection attacks	Prosumers	Electron volt exchange blockchain	Implementation of hyper ledger fabric	Bidding value and data from grid sensors	Distributed protocol of electron volt exchange has improved the privacy of the energy market	Contract on energy market was not considered
[257]	Increase the efficiency and security of energy market	Buyers and sellers of energy market	Decentralized blockchain on consortium ethereum network	Implementation of double auction with reduced blockchain overhead cost	Quantity, price and market identification number of consumption bid and generation bid	-	Competitive market based on prosumers has not considered
[258]	Minimize the electricity cost and flatten the load pattern	Residential, industrial and commercial	Blockchain	Decentralized identifier has been utilized as a digital identity for accessing blockchain	Availability of DG power and load profile, battery & fuelcell data of consumers	Digital identity has enhanced the system to identify the suspicious data	Real-time implementation requires the cloud computing and large system with huge number of participants
[259]	Minimize electrical energy consumption and maintain supply-demand balance	Smartgrid	Blockchain, IoT	Blockchain based distributed DSM	Data from smart meters, rewards or penalties and smart contract data	Privacy and anonymity has been improved with distributed environment	Prosumers have not been considered
[260]	Minimize the peak load and peak time stress on grid	Residential	Big data, and ML	ToU based DR	Tariff and electricity consumption data and the data collected from questionnaires	-	Impact of DGs and storage devices have not been considered
[261]	Minimize the sent data to electric power company with adaptive data compression	Residential	Big data, IoT with wireless sensor networks	Automatic energy management at consumer premises	Compressed data of demand and DG power availability as coefficients of cost function	-	Privacy and security issues on data compression and extraction have not been considered
[262]	Minimize the peak load	Residential, industrial and commercial consumers in DC microgrid	Blockchain	Implement power sharing between energy storage devices and consumers through bidirectional converters with fuzzy logic controller	Load demand, availability of power from energy storage device, tariffs and incentives	-	Privacy issues have not been considered and impact of uncertainites on demand have not considered
[263]	Minimize the electricity cost by scheduling the demand on most economical generating unit	Community microgrid	Blockchain	Blockchain allows the autonomous monitoring and billing through smart contracts	Historical data of demand and DG power for forecasting	-	Controllable appliances are much limited
[264]	Minimize the non-technical losses	Distribution system	Data mining	Data mining has been utilized to detect the non-technical losses	Power consumption data of individual consumers connected in the distribution network	-	Consumers with large range of consumption alone were considered which affects the accuracy of developed model
[265]	Minimize the energy consumption, cost & peak load and improve power quality	Residential consumers of islanded village with solar microgrid	Data logger	Data logger records the data and communicated it to the laptop or personal computer	Load pattern, PV power availability and battery data	-	The increasing the village size and its total consumption creates major impact in the reliability of the system

(continued on next page)

Renewable and Sustainable Energy Reviews 156 (2022) 111915

Table 8	3 (continued).						
Ref no	Objective	Consumers	Technology used	Contribution of technology	Data	Privacy issues	Inference
[266]	Minimize the load shedding or blockout by load shifting	Nanogrid	Cloud storage	The undervoltage and overvoltage relays have sent the load shedding signal to the consumers incase of extremely high peak loads. The consumers shift the loads to off-peak hour to avoid blackout	Voltage, frequency, power generation and load demand	The developed model affects the privacy of consumer as the information on the load pattern has been shared in cloud storage	Privacy and user comfort has been neglected
[267]	Maximize the profitability & consumer satisfaction and minimize the environmental impacts	Residential consumers	Blockchain	Public blockchain allows the consumers to verify or add the data which are validated by public key cryptography.	Dynamic pricing, DR signal, Energy profile, new schedule	Public blockchain with distributed control has provided limited privacy on data access to registered participants	Appliance scheduling not considered. Also DGs on consumer premises not included. Accessibility of data editing affects the security
[268]	Improve the accuracy of big data in DR	Residential consumers	Big data	DR has been implemented based on time and price	Population, appliances, living habits, income distribution and point of view on energy conservation of household	Privacy issues has existed due to sharing of huge information of household	The privacy and security issues have not been addressed
[269]	Minimize the peak load	Smartgrid	Big data	National virtual power plant platform enhanced with big data monitor and record the energy consumption data for automated DR	Power availability, demand, energy loss	-	The developed model has not been application specific and privacy & security issues has not been discussed
[270]	Minimize the peak load and improve the cost savings	Both small and large scale consumers	Big data	It has been utilized to minimize the peak load than forecasted peak load	Smart meter data and utility data	-	Privacy and security on data have not been considered. Computational complexity increases with increase in the size of system
[271]	Maximize the social welfare	Residential, commercial and industrial	Blockchain	Energy trading has been scheduled based on iterative double auction	Electricity consumption, solar power generation and trading cost	_	Privacy issues not addressed. Also, the scheduling of appliances not included
[272]	Minimize the cost and peak load	Small and medium range of consumers	Ethereum private blockchain	Private blockchain carried out the double auction without minimum data transfer through public blockchain	Power generation, demand, bidding values, final market price	Restricted data to the public blockchain has enhanced private blockchain	Transmission capacity and voltage constraints were not considered
[273]	Minimize the processing speed of big data	-	Big data	Fuzzy C-mean clustering has been used in power big data to fasten the DSM in daily load curve	Load pattern of consumers, Peak & off-peak durations, ToU tariff	-	Application specific issues has not been addressed
[274]	Forecast the load and generation using ML prediction model for DSM	Residential	Data mining	Based on the historical data from data mining, the load and generation has been forecasted with minimum mean squared error	Historical data of load pattern and power generation	-	Privacy issues were not addressed. HVAC loads alone were considered
[275]	Improve the security of DSM implementation in industrial IoT	Industrial and residential	Big data with IoT	Payload authentication framework with constrained application protocol improved the security and reliability of datafile	Load pattern	-	Third party tool known as Apache Spark has been used for data processing which affects the privacy of the consumer
[276]	Minimize the PAR	Consumers with EV	Big data and cloud computing	Data collection and processing for the implementation of DSM	Peak, off-peak durations & tariff and EV charging/discharging schedule	_	Load scheduling has not considered.
[277]	Minimize the electricity cost by remote monitoring and control of smart appliances	Smart home	Edge analytics	Embedded flexible edge analytics has been utilized to monitor and shift the electrical appliances operation from peak hour to off-peak hour	Data from sensors in the appliances, load pattern, status of actuators and controllers, power availability and battery status	_	Impact of DG in smart home were not considered. Complete load profile has been recorded but the privacy and security issues has not been addressed
[278]	Improve the preventive maintenance of DSM	Industrial, commercial and residential	Cloud analytics	Load shedding during peak duration has been efficiently prevented with historical data	Load pattern and peak load duration	_	Uncertainities and forecasting has not been considered.

Advanced metering infrastructure for DSM implementation.

ind ranced i	netering minubilitetare for Dom i	mpicinicination				
Ref No	Objective	Description	Technology/model	Communication	Advantages	Limitations
[288]	Reducing the deployment cost and prolonging network lifetime	Third generation partnership project technology has been utilized for communication with AMI which has reduced the cost and increase the lifetime	Narrow band IoT	Low power wide area network	With the existing communication networks, it is more robust and covered larger areas without high investment	Narrow band IoT utilizes cellular communication which is prone to natural and man-made disasters. It requires skilled personnel
[289]	Protection of information from cyber attacks	Light weight security solution has been suggested to communicate finger-print signal along with actual data	Blockchain	Low power wireless network	Secure and privacy enhanced data transfer and handling with no additional hardware requirement	Data provenance have tracked the origin of data rather than in between modifications. Therefore, data lineage might be a good alternate for Data provenance which tracks the flow of data from origin to the destination
[290]	Maximization of sum-rate to improve the communication efficiency	Cellular assisted device-to-device communication has been utilized for smart meters to cluster head. Wired communication has been used between cluster heads and meter data management unit	-	Cellular assisted device-to-device communication	The sum-rate has been improved by 25.3% than random algorithm.	The sum-rate has been reduced by 12.3% than linear random search
[28]	Improving the anonymity and untraceability of security in data transfer of AMI	Secure communication protocol has been developed based on random oracle model for data transfer between smart meters, AMI head end system and trusted authority	IoT	-	The security, anonymity, untraceability and performance of communication for AMI has been improved by random oracle model	Bits required for participant's identity, public/private keys, shared private keys, for hash values and time stamps have been assumed. Hence, the error due to the mismatch of assumed bit size and actual requirement affects the efficiency and performance of the model
[291]	Minimization of network traffic and improving the network performance	PLC has been utilized for the data transfer between the data concentrator and data storage	-	PLC	PLC has minimized the traffic on the network by 285.26 bits per second with negligible installation cost for communication link	Transmission lines for communication purpose injects noise signals on the data to be transferred. Therefore, it requires additional noise suppression devices.
[292]	Minimum cost of investment for AMI communication network	Cellular network has been utilized for the communication in AMI. The traffic on cellular network due to smart meters have been reduced by optimal choice of cellular base station with low traffic.	Scalable route map for wireless heterogeneous networks	Long term evolution cellular communication	Smart meters has communicated with the universal data aggregation points using existing cellular network	Utilization of the same network for mobile phone and smart meter is a short term planning. For long term usage, it requires the installation of additional base stations for network communication.
[293]	Detecting the fraudulent electricity access	Tunable fraud detection system has been utilized in AMI for detecting the ongoing fraud with short lived patterns. The optimization model have included the true or false alarms as a constraint. It has also included the maximization of revenue as an objective.	Short lived pattern model using fuzzy	-	The performance of short lived pattern has been proved to be efficient in detecting the fraud activities within short duration of time	Short lived patterns have designed the consumption pattern based on fewer sets of recent consumption data which may involve forecasting errors. Therefore, fault detection is a major limitation
[294]	Preventing the voltage collapse in the power system	When the SCADA detects any contingency activities in the power system, the event driven emergency DR program is initialized. The DR program is modeled as multi-objective optimization which controls the change in load pattern	SCADA	-	SCADA has centralized monitoring and control abilities which makes the process to implement in simple.	The centralized DR program implementation has a limitation that the process of optimal scheduling is complex due to the large network
[295]	Preventing the inference of privacy issues in sensitive loads	Privacy-utility trade-off algorithm has been developed for the secure data transfer between the smart meters and electrical utility companies			The developed algorithm has improved the privacy of smart meter data and insecure operation of sensitive appliances and permit the utility to track the load pattern of the smart home periodically.	The secure communication link has a major impact on the privacy of data transfer.
[296]	Improving the reliability and performance of communication in AMI	Grid topological routing has been utilized to select the optimal path for the data transfer between the smart meters and cluster heads	Grid topological router	-	The fault tolerance of developed routing model has been proved to be better than routing model for low-power & lousy networks and Ad-hoc networks	The response of grid topological routing during the uncertainities in network are less fault tolerant
[297]	Improving the computational efficiency and secure communication	The identity document-based authentication has been utilized for secure communication and access of power consumption data.	Identity document-based authentication	Neighborhood area network and wide area network	The developed model has enhanced the privacy on data between AMI and distributed system operator from cyber attacks.	The complexity of the model increases the size of the network

(continued on next page)

Table 9 (continued).

Table 9 (c										
Ref No	Objective	Description	Technology/model	Communication	Advantages	Limitations				
[298]	Capacity estimation of network for resource allocation	Consumers having similar consumption pattern has formed into clusters and the consumption pattern for each cluster has optimally scheduled and communicated to the supplier	Data mining	_	The overall consumption at any time slot or peak load of the cluster has been reduced based on the developed model	The effectiveness of the developed model on load reduction is very minimum which requires additional load shifting action				
[299]	Minimization of electricity cost, PAR, user discomfort and carbon dioxide emission	Hybrid bacterial foraging and particle swarm optimization (HBFPSO) algorithm has been utilized in energy management controller to achieve the objectives. It has implemented ToU tariff.	-	Wireless local area network	The developed model could be applied for residential, industrial and commercial sectors	RTP tariff has not been considered. Temperature dependent loads in all three sectors has not been scheduled by considering temperature as a constraint				
[300]	Minimization of outage duration	The outage management system follows the steps of fault identification, fault location, consumer notification and service restoration	-	Narrow band and broad band PLC	Additional investment on communication lines has been avoided. Notification of outages to the consumer could enhance the energy management of consumers	PLC requires noise suppression to avoid the false alarm of outage to consumer and it has utilized 597 bytes for communication of data which is very large				
[301]	Minimize the electricity cost	Collaborative filtering algorithm has been utilized for determining the most suitable tariff to minimize the cost	Cloud storage	Wide area network and home area network	Individual user has the choice of selecting the tariff	Optimal scheduling of load demand has not been considered				
[284]	Monitor the electricity consumption and power quality	Adaptive data compression has been used to reduce the data size	ΙσΤ	Local area network	It supports the implementation of DSM and DSM could enhance the power quality	The DGs and storage devices has not been included. Also, dynamic pricing based DR has not been considered				
[302]	Monitoring the DG power, building and home energy management	Solar PV and wind power data has been transferred and stored through Zigbee for DSM implementation	Zigbee	Wireless service network	Zigbee devices has been proved to be sufficient for high rise buildings and several urban homes. The developed energy management model has allocated the most economical source	Appliance scheduling has not been considered				
[303]	Maximize energy savings and user comfort	Appliances, smart sensors and energy technologies have been monitored and controlled through Zigbee	Zigbee	Wireless service network	Ubiquitous home networks with zigbee devices are flexible, easy to implement and requires minimum cost	Limited number of appliances were considered and impact of DGs have not been included. Air conditioners should be modeled based on temperature				
[304]	Minimization of peak load and distribution losses	The developed DSM model has evident to provide uninterruptible power supply to residential consumer during peak hour		Local area network and wide area network	The reliability of the system has improved	The developed model has concentrated on resource allocation than load pattern change				

26



Fig. 8. Impacts of DSM on PQ.

harmonics. If the large resistive electric heating loads have disconnected, it would have reduced the damping which provides higher distortion levels. The damping can be increased by adding either the resistive part of passive filters or by means of the active filters. The impacts of micro-generation and the DSM on PQ (voltage fluctuation and unbalance) have been analyzed in [312] with a case study in a low voltage radial distribution network. It was proved that the voltage variations could not reach beyond the threshold values when the DSM strategy implemented in the system. Section 7 provides the research gap of the DSM implementation in the microgrid for future research. The DSM implementation could be utilized for improving the stability and reliability of the microgrid. The DSM has been achieved voltage control, frequency and the peak load reduction in [313]. In [314], reliability and voltage stability has been suggested to include while implementing the DSM programs. The different PQ issues such as voltage and frequency variations, reliability, stability and their improvement have been critically discussed in Table 10.

7. Future scope

DSM implementation in the microgrid has become popular in the last two decades but a few areas like energy storage, PQ issues, environmental issues, forecasting error, implementation of the latest technologies should be focused on for further development. The utilization of a hybrid storage system (both battery and EV) for the enhancement of BESS whereas the installation of a distributed ESS in the microgrid could be the best solution for the effective utilization of BESS. Fuel cellbased EVs could be utilized as an emission-free transport and also a storage device for DSM. Mobile BESS could be considered as common storage for microgrids that having different usage patterns instead of using a static BESS. Also, it could be incorporated in microgrids where the battery was kept idle for a longer duration. The impact of uncertainties of DGs could be considered for the resource allocation. The forecasting errors could be minimized by analyzing the mean absolute percent error of the on-site generation. The most commonly used on-site generations were PV and wind. Other DGs without uncertainty issues such as CHP and biomass-based power plants need to be considered for efficient resource allocation and load shifting.

Frequent switching of loads in DSM causes some PQ issues such as harmonics, power factor, voltage deviation in the microgrid. Such parameters could be added as a constraint of the optimization problem along with the power balance, storage and other constraints in order to overcome the above-said drawbacks. The voltage spring concept could be included in the implementation of DSM to avoid the voltage unbalance caused by the frequent switching of loads. Load shifting in industrial consumers and commercial consumers is much limited than residential consumers. Hence, industrial consumers and shiftable commercial appliances could be scheduled to reduce the peak load of the utility.

Online energy management controllers could be used for load forecasting and sizing of BESS. Real-time implementation of load shifting in residential consumer needs further advancements such as secure data storage with privacy enhanced data transfer. The implementation of advanced metering infrastructure is required for effective DSM implementation. Internet of Energy, blockchain technology, ML and big data analytics could be used for the real-time implementation of DSM. Market-based analysis should be done to improve the tariff structure. The real-time implementation of the price-based DR program requires certain modifications such as the inclusion of power factor and harmonic contribution of the consumer. Peer-to-peer energy trading could be implemented for a microgrid with prosumers to achieve economic energy sharing among the prosumers with minimum transmission losses. Privacy and security issues related to the data transfer, storage and handling requirements should be increased attention in real-time DSM implementation.

8. Conclusion

The review article has been discussed the different types of microgrid layers with different DSM implementations, economic and environmental behaviors of the microgrid while the implementation of DSM. Also, the article has been critically discussed with the recent trends in the different types of DR programs such as the IBDR program and pricebased DR program. A detailed comparison has been made to select the appropriate DR technique for different applications, storage devices and DGs that are used in the microgrid. The overview of soft computing techniques has been discussed based on the mathematical model and optimization algorithms. The DSM implementation in India has been discussed with the ToD tariff for different states. The recent advancements in DSM and the impact of DSM in different PQ issues have been examined. As a result, this review article provides the useful insights to the future researchers who are seeking the DSM implementation in various aspects such as data science, energy storage and EV.

Impact of DS	SM on PQ, stability ar	nd power flow issues.				
Ref No	Key issues	Causes	DSM	Contribution	Benefits	Inference
[315]	PQ (Harmon- ics,Voltage unbalance, electromag- netic interference)	Electronic Adjustable Speed Drives (ASD)	Energy efficiency, peak demand reduction	Improvement in efficiency, minimizes the electricity consumption and cost	ASD with energy efficient motor have been used to obtain the cost savings in residential, commercial and industrial sectors	Investment on energy efficient motor is high and the respective cost minimization is very minimum
[316]	Voltage sag, harmonics	Insufficient var support, high short circuit currents, arc furnaces, ASD, PE devices	Energy conservation	Energy conservation measures reduce the total energy consumed during peak duration	Integration of power electronic controllers and communication with distribution automation to reduce the distribution and customer losses.	Energy conservation measures the consumer comfort and it is suitable only for industrial consumers
[317]	Voltage distortion, power factor, harmonics	Electronic ballast, ASD	Energy efficiency	Replacement of fluorescent lamp by CFL increases the energy savings	Impact of using CFLs for minimum energy consumption has achieved the harmonics mitigation and reduced the voltage distortion	It is not beneficial for consumers. Only utility get benefits in fuel savings
[318]	Voltage swell, harmonics	Grid integration of DG	On-site generation, DLC	On-site DG reduces the peak load	On-site generation has reduced the cost of transmission and distribution in addition to peak load reduction	Co-ordinated control is required to avoid the technical issues and strategies for implementing the DSM
[319]	Harmonics, power factor	Electronic ballast of fluorescent lamps	Energy efficiency	Replacement of conventional ballast with electronic ballast has reduced the energy consumption. The electronic ballast is modeled with high frequency resonant inverter	Power factor has improved and radio interference has reduced	High electromagnetic interference, high switching losses, large size and heavy weight
[320]	Reliability and uncertainity	Intermittency and uncertainity of DG	RTP	Optimal sizing of energy storage system with implementation of RTP-based DR has reduced the peak load and electricity cost	Microgrid with DG, DSM and sizing of ESS reduces the cost, improve the reliability by reducing the outages	Only reliability and uncertainity issues has addressed. Stability of the microgrid due to the renewable energy penetration was not considered
[321]	Reliability	Poor load pattern with high peaks and deep valleys	Peak shifting	Impact of DSM on the system reliability with operating consideration model.	Included the duty cycle, operating reserve policy, outage postponability and unit commitment policy as the constraints which enhanced reliability of the system	Energy recovery and load diversity was not considered
[322]	Power flow (reactive power)	Power shortage during peak duration	Load shedding	DSM quality index and DSM appreciation index has been utilized for analyzing the techno-economical benefits of peak shifting	DSM quality index monitor the existence of reactive power whereas DSM appreciation index monitor the cost savings due to the implementation of DSM which has enabled the control of both power quality and cost minimization	The developed model is only applicable for industrial consumers.
[323]	Reliability	Load forecast uncertainity	Peak shifting	Impact of load forecast uncertainity on peak shifting has reduced by improving the reliability indices such as loss of load expectation and loss of energy expectation	Implementation of load shifting with reliability analysis has reduced the peak load	Generation forecasting uncertainity was not considered

(continued on next page)

D. Kanakadhurga and N. Prabaharan

Table 10 (continued).

Table 10 (co	munueu).					
Ref No	Key issues	Causes	DSM	Contribution	Benefits	Inference
[324]	Reliability	Power fluctuation of renewable energy generation	Peak shaving and valley filling	Flat load pattern has obtained by implementing DR programs based on availability of DG power and ESS	Reliability indices such as loss of load probability, loss of load expectation and expected energy not supplied has proved the effectiveness of DR program in achieving the flat load curve	Impact of DGs and their uncertainties were not considered
[313]	Stability	Load uncertainity	Frequency linked pricing	Storage devices at consumers premises has utilized to provide both active and reactive power support to distribution utility by paying the frequency linked pricing which improve the voltage stability	Real and reactive power support has improved the voltage stability and reduced the load shedding in the distribution network	Initial cost of investment is high (infrastructure cost). Success depends on the willingness of the consumer to participate in DSM program.
[325]	Power flow (Voltage and frequency control)	Uncertainity and intermittency of solar PV	Electric spring	Frequency and voltage control has achieved by injecting active power. d-q transformation has utilized for independent control of voltage & frequency at the distribution end.	Electric spring has enhanced the voltage and frequency control of microgrid to restore it to nominal rating within a short duration of time	Selectivity, zone of protection of relay affects the operation of electric spring were not considered
[326]	Stability	Uncertainity and intermittency of DG	On-site generation	PI controller with atom search optimization algorithm has utilized to maintain the stable with DSM implementation	BESS along with solar dish-stirling arrangement has reduced the peak load that consumed from utility and maintained the voltage stability	Computationally complex for larger microgrids
[327]	Power flow control	Fluctuations in frequency, voltage and power due to penetration of PV energy	On-site generation	On-site generation has reduced the peak demand and power fluctuations while including the battery and super capacitor bank	Droop control has been utilized for power flow control with grid integration of PV system and storage device	Transient performance with dynamic loads were not considered
[328]	Stability	Power shortage during peak duration	Adaptive load shedding and EDRP	Adaptive load shedding and EDRP has used to improve the stability of the system and minimize the frequency drop	Reduced power interruption and probability of failure in 39-bus distribution system has obtained by minimizing the convergence time and frequency drop	Load shedding affects the consumer comfort
[329]	Power flow control	Voltage instability due to high peak load	EDRP	Droop control with non sorting GA II has used for optimal power flow control with EDRP	Load factor has improved and Peak to valley difference has reduced	Transient performance with load dynamics were not considered. Flexibility of the optimal power flow with other DR programs were not addressed
[330]	Power flow control	Forecasting error of RES	RTP	Model predictive control has utilized to reduce the forecast errors.	Peak to valley difference, load fluctuation and total cost has reduced but the utilization of RES is increased to 100%	Impact of storage devices in uncertainity was not considered
[331]	Power flow control	Impact of load on radial and meshed distribution networks	Load shifting	Demand-responsive loads has used to maximize the network benefits by optimizing upstream network constraints and operating reserve	Minimized the electricity cost by maintaining the real and reactive power flow through load shifting from peak to off-peak hour	Operational constraints of load were not considered

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgment

The authors are thankful to SASTRA Deemed University, India for the support provided.

References

- [1] Theo Wai Lip, Lim Jeng Shiun, Ho Wai Shin, Hashim Haslenda, Lee Chew Tin. Review of distributed generation (DG) system planning and optimisation techniques: Comparison of numerical and mathematical modelling methods. Renew Sustain Energy Rev 2017;67:531–73.
- [2] Ela Erik, O'Malley Mark. Studying the variability and uncertainty impacts of variable generation at multiple timescales. IEEE Trans Power Syst 2012;27(3):1324–33.
- [3] Ela Erik, Kirby Brendan, Lannoye Eamonn, Milligan Michael, Flynn Damian, Zavadil Bob, et al. Evolution of operating reserve determination in wind power integration studies. In: IEEE PES Gen. Meet.; 2010, p. 1–8.
- [4] Tan Xingguo, Li Qingmin, Wang Hui. Advances and trends of energy storage technology in microgrid. Int J Electr Power Energy Syst 2013;44(1):179–91.
- [5] Majumder Ritwik. Modeling, stability analysis and control of microgrid (Ph.D. thesis), (February):Queensland university of technology; 2010, p. 182.
- [6] Marnay Chris, Chatzivasileiadis Spyros, Abbey Chad, Iravani Reza, Joos Geza, Lombardi Pio, et al. Microgrid evolution roadmap. In: 2015 International symposium on smart electric distribution systems and technologies. IEEE; 2015, p. 139–44.
- [7] García Vera Yimy E, Dufo-López Rodolfo, Bernal-Agustín José L. Energy management in microgrids with renewable energy sources: A literature review. Appl Sci 2019;9(18).
- [8] Katiraei F, Iravani MR, Lehn PW. Micro-grid autonomous operation during and subsequent to islanding process. IEEE Trans Power Deliv 2005;20(1):248–57.
- [9] Shafiee Qobad, Stefanovic Cedomir, Dragicevic Tomislav, Popovski Petar, Vasquez Juan Carlos, Guerrero Josep M. Robust networked control scheme for distributed secondary control of islanded MicroGrids. IEEE Trans Ind Electron 2014;61(10):5363–74.
- [10] Wang Ran, Wang Ping, Xiao Gaoxi, Gong Shimin. Power demand and supply management in microgrids with uncertainties of renewable energies. Int J Electr Power Energy Syst 2014;63:260–9.
- [11] Xu Y, Li Z. Distributed optimal resource management based on the consensus algorithm in a microgrid. IEEE Trans Ind Electron 2015;62(4):2584–92.
- [12] Vos Arthur. Effective business models for demand response under the Smart Grid paradigm. In: 2009 IEEE/PES power syst. conf. expo.; 2009, p. 1.
- [13] Government of India, Central Eectricity Authority Ministry of power. All India installed capacity (in MW) of power stations (As on 31-01-2021), Vol. 4. Technical report, Cent. Eectricity Authority, Minist. Power; 2021, p. 1–7.
- [14] The Energy and Resources Institute. Renewable energy and green growth in India. Technical report, 2019, p. 1–26.
- [15] Norton John W. Decentralized systems. In: Water environ. Res. Vol. 81, no. 10. 2009, p. 1440–50.
- [16] Jiayi Huang, Chuanwen Jiang, Rong Xu. A review on distributed energy resources and MicroGrid. Renew Sustain Energy Rev 2008;12(9):2472–83.
- [17] Basak Prasenjit, Chowdhury S, nee Dey S Halder, Chowdhury SP. A literature review on integration of distributed energy resources in the perspective of control, protection and stability of microgrid. Renew Sustain Energy Rev 2012;16(8):5545–56.
- [18] Brearley Belwin J, Prabu R Raja. A review on issues and approaches for microgrid protection. Renew Sustain Energy Rev 2017;67:988–97.
- [19] Kaur Amandeep, Kaushal Jitender, Basak Prasenjit. A review on microgrid central controller. Renew Sustain Energy Rev 2016;55:338–45.
- [20] Rajesh KS, Dash SS, Rajagopal Ragam, Sridhar R. A review on control of ac microgrid. Renew Sustain Energy Rev 2017;71:814–9.
- [21] Alasseri Rajeev, Tripathi Ashish, Joji Rao T, Sreekanth KJ. A review on implementation strategies for demand side management (DSM) in Kuwait through incentive-based demand response programs. Renew Sustain Energy Rev 2017;77:617–35.
- [22] Ming Zeng, Song Xue, Mingjuan Ma, Lingyun Li, Min Cheng, Yuejin Wang. Historical review of demand side management in China: Management content, operation mode, results assessment and relative incentives. Renew Sustain Energy Rev 2013;25:470–82.
- [23] Warren Peter. A review of demand-side management policy in the UK. Renew Sustain Energy Rev 2014;29:941–51.

- [24] Pallonetto Fabiano, De Rosa Mattia, D'Ettorre Francesco, Finn Donal P. On the assessment and control optimisation of demand response programs in residential buildings. Renew Sustain Energy Rev 2020;127:109861.
- [25] Behrangrad Mahdi. A review of demand side management business models in the electricity market. Renew Sustain Energy Rev 2015;47:270–83.
- [26] Meyabadi A Fattahi, Deihimi MH. A review of demand-side management: Reconsidering theoretical framework. Renew Sustain Energy Rev 2017;80:367–79.
- [27] Uddin Moslem, Romlie Mohd Fakhizan, Abdullah Mohd Faris, Abd Halim Syahirah, Abu Bakar Ab Halim, Chia Kwang Tan. A review on peak load shaving strategies. Renew Sustain Energy Rev 2018;82:3323–32.
- [28] Mahmood Khalid, Arshad Jehangir, Chaudhry Shehzad Ashraf, Kumari Saru. An enhanced anonymous identity-based key agreement protocol for smart grid advanced metering infrastructure. Int J Commun Syst 2019;32(16):e4137.
- [29] Vázquez-Canteli José R, Nagy Zoltán. Reinforcement learning for demand response: A review of algorithms and modeling techniques. Appl Energy 2019;235:1072–89.
- [30] Pham Quoc-Viet, Liyanage Madhusanka, Deepa N, VVSS Mounik, Reddy Shivani, Maddikunta Praveen Kumar Reddy, et al. Deep learning for intelligent demand response and smart grids: A comprehensive survey. 2021, arXiv preprint arXiv:2101.08013.
- [31] Pallonetto Fabiano, De Rosa Mattia, Milano Federico, Finn Donal P. Demand response algorithms for smart-grid ready residential buildings using machine learning models. Appl Energy 2019;239:1265–82.
- [32] Meeuw Arne, Schopfer Sandro, Wörner Anselma, Tiefenbeck Verena, Ableitner Liliane, Fleisch Elgar, et al. Implementing a blockchain-based local energy market: Insights on communication and scalability. Comput Commun 2020;160:158–71.
- [33] Hirsch Adam, Parag Yael, Guerrero Josep. Microgrids: A review of technologies, key drivers, and outstanding issues. Renew Sustain Energy Rev 2018;90:402–11.
- [34] Martin-Martínez F, Sánchez-Miralles A, Rivier M. A literature review of microgrids: A functional layer based classification. Renew Sustain Energy Rev 2016;62:1133–53.
- [35] Gottschalk Marion, Uslar Mathias, Delfs Christina. The Use case and smart grid architecture model approach: the IEC 62559-2 use case template and the SGAM applied in various domains. Springer; 2017.
- [36] Estebsari Abouzar, Barbierato Luca, Bahmanyar Alireza, Bottaccioli Lorenzo, Macii Enrico, Patti Edoardo. A SGAM-based test platform to develop a scheme for wide area measurement-free monitoring of smart grids under high PV penetration. Energies 2019;12(8):1417.
- [37] Gottschalk Marion, Uslar Mathias, Delfs Christina. The smart grid architecture model–SGAM. In: The use case and smart grid architecture model approach. Springer; 2017, p. 41–61.
- [38] Uslar Mathias, Rohjans Sebastian, Neureiter Christian, Pröstl Andrén Filip, Velasquez Jorge, Steinbrink Cornelius, et al. Applying the smart grid architecture model for designing and validating system-of-systems in the power and energy domain: A European perspective. Energies 2019;12(2):258.
- [39] Nayar CV, Phillips SJ, James WL, Pryor TL, Remmer D. Novel wind/diesel/battery hybrid energy system. Sol Energy 1993;51(1):65–78.
- [40] EPA- Environmental Protection Agency. Distributed generation-electricity and its environmental impacts. In: Energy environ., 2018.
- [41] Jenkins N, Ekanayake JB, Strbac G. Distributed generation. Distrib Gener 2010;1–279.
- [42] Salas V, Olías E. Overview of the state of technique for PV inverters used in low voltage grid-connected PV systems: Inverters above 10 kW. Renew Sustain Energy Rev 2011;15(2):1250–7.
- [43] Carpinelli Guido, Mottola Fabio, Proto Daniela. Optimal scheduling of a microgrid with demand response resources. IET Gener Trans Distrib 2014;8(12):1891–9.
- [44] Adika Christopher O, Wang Lingfeng. Autonomous appliance scheduling for household energy management. IEEE Trans Smart Grid 2014;5(2):673–82.
- [45] Javaid Nadeem, Ullah Ihsan, Akbar Mariam, Iqbal Zafar, Khan Farman Ali, Alrajeh Nabil, et al. An intelligent load management system with renewable energy integration for smart homes. IEEE Access 2017;5:13587–600.
- [46] Celik Berk, Roche Robin, Bouquain David, Miraoui Abdellatif. Decentralized neighborhood energy management with coordinated smart home energy sharing. IEEE Trans Smart Grid 2018;9(6):6387–97.
- [47] Mediwaththe Chathurika P, Shaw Marnie, Halgamuge Saman, Smith David B, Scott Paul. An incentive-compatible energy trading framework for neighborhood area networks with shared energy storage. IEEE Trans Sustain Energy 2020;11(1):467–76.
- [48] Mahmud Khizir, Hossain M Jahangir, Town Graham E. Peak-load reduction by coordinated response of photovoltaics, battery storage, and electric vehicles. IEEE Access 2018;6:29353–65.
- [49] Wiki Distributed Wind. Distributed wind Wiki. U.S. Dep. Energy; 2018, p. 1–5.
- [50] Arifujjaman Md, Iqbal M Tariq, Quaicoe John E. Energy capture by a small wind-energy conversion system. Appl Energy 2008;85(1):41–51.
- [51] Moura Pedro S, de Almeida Aníbal T. The role of demand-side management in the grid integration of wind power. Appl Energy 2010;87(8):2581–8.

- [52] Aghajani GR, Shayanfar HA, Shayeghi H. Demand side management in a smart micro-grid in the presence of renewable generation and demand response. Energy 2017;126:622–37.
- [53] Wang Jianhui, Liu Cong, Ton Dan, Zhou Yan, Kim Jinho, Vyas Anantray. Impact of plug-in hybrid electric vehicles on power systems with demand response and wind power. Energy Policy 2011;39(7):4016–21.
- [54] Wu Chenye, Mohsenian-Rad Hamed, Huang Jianwei, Wang Amy Yuexuan. Demand side management for wind power integration in microgrid using dynamic potential game theory. In: 2011 IEEE GLOBECOM workshops. IEEE; 2011, p. 1199–204.
- [55] Alham MH, Elshahed M, Ibrahim Doaa Khalil, El Zahab Essam El Din Abo. A dynamic economic emission dispatch considering wind power uncertainty incorporating energy storage system and demand side management. Renew Energy 2016;96:800–11.
- [56] Kumar P Satish, Chandrasena RPS, Ramu V, Srinivas GN, Babu K Victor Sam Moses. Energy management system for small scale hybrid wind solar battery based microgrid. IEEE Access 2020;8:8336–45.
- [57] Energy education Electrical load [Online].
- [58] Gellings Clark W. The concept of demand-side management for electric utilities. Proc IEEE 1985;73(10):1468–70.
- [59] Li Bosong, Shen Jingshuang, Wang Xu, Jiang Chuanwen. From controllable loads to generalized demand-side resources: A review on developments of demand-side resources. Renew Sustain Energy Rev 2016;53:936–44.
- [60] Dehghan-Dehnavi Somayeh, Fotuhi-Firuzabad Mahmud, Moeini-Aghtaie Moein, Dehghanian Payman, Wang Fei. Estimating participation abilities of industrial customers in demand response programs: A two-level decision-making tree analysis. In: 2020 IEEE/IAS 56th industrial and commercial power systems technical conference. IEEE; 2020, p. 1–8.
- [61] Hemanth GR, Charles Raja S, Jeslin Drusila Nesamalar J, Senthil Kumar J. Cost effective energy consumption in a residential building by implementing demand side management in the presence of different classes of power loads. Adv Build Energy Res 2020;1–26.
- [62] Conejo Antonio J, Morales Juan M, Baringo Luis. Real-time demand response model. IEEE Trans Smart Grid 2010;1(3):236–42.
- [63] Deshmukh Ranjit, Ghatikar Girish, Yin Rongxin, Das G Ganesh, Saha Sujay Kumar. Estimation of potential and value of demand response for industrial and commercial consumers in Delhi. 2015.
- [64] Jiang Xunyan, Wu Lei. A residential load scheduling based on cost efficiency and consumer's preference for demand response in smart grid. Electr Power Syst Res 2020;186:106410.
- [65] Jiang Xunyan, Wu Lei. Residential power scheduling based on cost efficiency for demand response in smart grid. IEEE Access 2020;8:197324–36.
- [66] Merdanoğlu Hakan, Yakıcı Ertan, Doğan O Tufan, Duran Serhan, Karatas Mumtaz. Finding optimal schedules in a home energy management system. Electr Power Syst Res 2020;182:106229.
- [67] Waseem Muhammad, Lin Zhenzhi, Liu Shengyuan, Sajjad Intisar Ali, Aziz Tarique. Optimal GWCSO-based home appliances scheduling for demand response considering end-users comfort. Electr Power Syst Res 2020;187:106477.
- [68] Wang Xiuwang, Mao Xinna, Khodaei Hossein. A multi-objective home energy management system based on internet of things and optimization algorithms. J Build Eng 2021;33:101603.
- [69] Duman A Can, Erden Hamza Salih, Gönül Ömer, Güler Önder. A home energy management system with an integrated smart thermostat for demand response in smart grids. Sustainable Cities Soc 2021;65:102639.
- [70] Hakimi Seyed Mehdi, Hasankhani Arezoo. Intelligent energy management in off-grid smart buildings with energy interaction. J Cleaner Prod 2020;244:118906.
- [71] Javadi Mohammad, Nezhad Ali Esmaeel, Firouzi Kimia, Besanjideh Farnoosh, Gough Matthew, Lotfi Mohamed, et al. Optimal operation of home energy management systems in the presence of the inverter-based heating, ventilation and air conditioning system. In: 2020 IEEE international conference on environment and electrical engineering and 2020 IEEE industrial and commercial power systems Europe. IEEE; 2020, p. 1–6.
- [72] Kuzlu Murat, Pipattanasomporn Manisa, Rahman Saifur. Communication network requirements for major smart grid applications in HAN, NAN and WAN. Comput Netw 2014;74–88.
- [73] Gungor V Cagri, Sahin Dilan, Kocak Taskin, Ergut Salih, Buccella Concettina, Cecati Carlo, et al. A survey on smart grid potential applications and communication requirements. IEEE Trans Ind Inform 2013;9(1):28–42.
- [74] Gao Jingcheng, Xiao Yang, Liu Jing, Liang Wei, Chen C LPhilip. A survey of communication/networking in smart grids. Future Gener Comput Syst 2012;28(2):391–404.
- [75] Nafi Nazmus S, Ahmed Khandakar, Gregory Mark A, Datta Manoj. Software defined neighborhood area network for smart grid applications. Future Gener Comput Syst 2018;79:500–13.
- [76] Ancillotti Emilio, Bruno Raffaele, Conti Marco. The role of communication systems in smart grids: Architectures, technical solutions and research challenges. Comput Commun 2013;36(17–18):1665–97.

- [77] Pattanaik P Akash, Sahoo N C, Mishra S. Implementation of demand side management using microcontroller and wireless communication. In: Proc. 2017 2nd IEEE int. conf. electr. comput commun technol.; 2017.
- [78] Rinaldi Stefano, Pasetti Marco, Sisinni Emiliano, Bonafini Federico, Ferrari Paolo, Rizzi Mattia, et al. On the mobile communication requirements for the demand-side management of electric vehicles. Energies 2018;11(5).
- [79] Luo Fengji, Ranzi Gianluca, Wang Shu, Dong Zhao Yang. Hierarchical energy management system for home microgrids. IEEE Trans Smart Grid 2018;10(5):5536–46.
- [80] Sen Sachidananda, Kumar Vishal. Microgrid control: A comprehensive survey. Annu Rev Control 2018;45:118–51.
- [81] Tayab Usman Bashir, Roslan Mohd Azrik Bin, Hwai Leong Jenn, Kashif Muhammad. A review of droop control techniques for microgrid. Renew Sustain Energy Rev 2017;76:717–27.
- [82] Akbari M, Golkar MA, Tafreshi SMM. Voltage control of a hybrid ac/dc microgrid in grid-connected operation mode. In: 2011 IEEE PES int. conf. innov. smart grid technol.. IEEE; 2011, p. 358–62.
- [83] Adhikari Sarina, Li Fangxing. Coordinated V-f and P-Q control of solar photovoltaic generators with MPPT and battery storage in microgrids. IEEE Trans Smart Grid 2014;5(3):1270–81.
- [84] Kostková K, Omelina Ĺ, Kyčina P, Jamrich P. An introduction to load management. Electr Power Syst Res 2013;95:184–91.
- [85] Gyamfi Samuel, Krumdieck Susan, Urmee Tania. Residential peak electricity demand response - highlights of some behavioural issues. Renew Sustain Energy Rev 2013;25:71–7.
- [86] Hamidi Vandad, Li Furong, Robinson Francis. Demand response in the UK's domestic sector. Electr Power Syst Res 2009;79(12):1722–6.
- [87] Wang Lingfeng, Wang Zhu, Yang Rui. Intelligent multiagent control system for energy and comfort management in smart and sustainable buildings. IEEE Trans Smart Grid 2012;3(2):605–17.
- [88] Tsikalakis Antonis G, Hatziargyriou Nikos D. Centralized control for optimizing microgrids operation. In: IEEE power energy soc. gen. meet.. IEEE; 2011, p. 1–8.
- [89] Ravindra Kumudhini, Iyer Parameshwar P. Decentralized demand-supply matching using community microgrids and consumer demand response: A scenario analysis. Energy 2014;76:32–41.
- [90] Javaid Nadeem, Hafeez Ghulam, Iqbal Sohail, Alrajeh Nabil, Alabed Mohamad Souheil, Guizani Mohsen. Energy efficient integration of renewable energy sources in the smart grid for demand side management. IEEE Access 2018;6:77077–96.
- [91] Li Chaojie, Yu Xinghuo, Yu Wenwu, Chen Guo, Wang Jianhui. Efficient computation for sparse load shifting in demand side management. IEEE Trans Smart Grid 2017;8(1):250–61.
- [92] Pourbabak Hajir, Chen Tao, Su Wencong. Centralized, decentralized, and distributed control for energy internet. In: The energy internet. Elsevier; 2019, p. 3–19.
- [93] Hecht Christian, Sprake David, Vagapov Yuriy, Anuchin Alecksey. Domestic demand-side management: analysis of microgrid with renewable energy sources using historical load data. Electr Eng 2021;1–16.
- [94] Rieger Alexander, Thummert Robert, Fridgen Gilbert, Kahlen Micha, Ketter Wolfgang. Estimating the benefits of cooperation in a residential microgrid: A data-driven approach. Appl Energy 2016;180:130–41.
- [95] Fridgen Gilbert, Kahlen Micha, Ketter Wolfgang, Rieger Alexander, Thimmel Markus. One rate does not fit all: An empirical analysis of electricity tariffs for residential microgrids. Appl Energy 2018;210:800–14.
- [96] Sanz JF, Matute G, Bludszuweit H, Laporta E. Microgrids, a new business model for the energy market. Renew Energy Power Qual J 2014;1(12):868–73.
- [97] Eto Joseph. The past, present, and future of U.S. utility demand side management programs. Rev L'Energie 1996;47(483):626–38.
- [98] Wayne Mary. Demand planning in the '80s. EPRI J 1984;9(10):6-15.
- [99] Sharma Ankit Kumar, Saxena Akash. A demand side management control strategy using whale optimization algorithm. SN Appl Sci 2019;1(8):1–15.
- [100] Darley John M. Energy conservation techniques as innovations, and their diffusion. Energy Build 1978;1(3):339-43.
- [101] Garg A, Maheshwari J, Mahapatra D. Energy conservation and commercialization in Gujarat. Technical report, 2009.
- [102] Jindal Anish, Singh Mukesh, Kumar Neeraj. Consumption-aware data analytical demand response scheme for peak load reduction in smart grid. IEEE Trans Ind Electron 2018;65(11):8993–9004.
- [103] Kumar K Prakash, Saravanan B. Day ahead scheduling of generation and storage in a microgrid considering demand side management. J Energy Storage 2019;21:78–86.
- [104] Sedighizadeh Mostafa, Mohammadpour Amir Hosein, Alavi Seyed Mohammad Mahdi. A two-stage optimal energy management by using ADP and HBB-BC algorithms for microgrids with renewable energy sources and storages. J Energy Storage 2019;21:460–80.
- [105] Gholami Amin, Ansari Javad, Jamei Mahdi, Kazemi Ahad. Environmental/economic dispatch incorporating renewable energy sources and plug-in vehicles. IET Gener Trans Distrib 2014;8(12):2183–98.
- [106] Burke Kerry B. The reliability of distributed solar in critical peak demand: A capital value assessment. Renew Energy 2014;68:103–10.

- [107] Gabriel Mel, Nathwani Jatin. Meeting the texas electricity peak demand conundrum: A case for wind and solar. Electr J 2018;31(1):57–64.
- [108] Zhang Shuo, Xiong Rui. HESS And its application in series hybrid electric vehicles. In: Model. dyn. control electrified veh.. 2018, p. 77–119.
- [109] Liang Huijun, Liu Yungang, Li Fengzhong, Shen Yanjun. Dynamic economic/emission dispatch including PEVs for peak shaving and valley filling. IEEE Trans Ind Electron 2019;66(4):2880–90.
- [110] Tushar Mosaddek Hossain Kamal, Zeineddine Adel W, Assi Chadi. Demandside management by regulating charging and discharging of the EV, ESS, and utilizing renewable energy. IEEE Trans Ind Inform 2018;14(1):117–26.
- [111] Saini S. Demand-side management module. In: Sustain. energy regul. policymaking Africa. 2007, p. 1–100.
- [112] Breukers SC, Heiskanen E, Brohmann B, Mourik RM, Feenstra CFJ. Connecting research to practice to improve energy demand-side management (DSM). Energy 2011;36(4):2176–85.
- [113] Shen Jingshuang, Jiang Chuanwen, Liu Yangyang, Qian Jie. A microgrid energy management system with demand response for providing grid peak shaving. Electr Power Compon Syst 2016;44(8):843–52.
- [114] Zhou Qin, Hou Fang, Huang Yuteng. Customer profiling-based optimal load shaving solution. In: Proc. 2013 IEEE 7th int. power eng. optim. Conf.; 2013, p. 250–5.
- [115] Wang Jun, Raza Ashhar, Hong Tianqi, Sullberg Adriana Cisco, De León Francisco, Huang Qi. Analysis of energy savings of CVR including refrigeration loads in distribution systems. IEEE Trans Power Deliv 2018;33(1):158–68.
- [116] Li Fucun, Guo Hongxia, Jing Zhen, Wang Zhaojun, Wang Xinku. Peak and valley regulation of distribution network with electric vehicles. J Eng 2019;2019(16):2488–92.
- [117] Shen Jingshuang, Jiang Chuanwen, Liu Yangyang, Qian Jie. A microgrid energy management system with demand response for providing grid peak shaving. Electr Power Compon Syst 2016;44(8):843–52.
- [118] Sanjeev P, Padhy Narayana Prasad, Agarwal Pramod. Peak energy management using renewable integrated DC microgrid. IEEE Trans Smart Grid 2018;9(5):4906–17.
- [119] Parizy Ehsan Saeidpour, Bahrami Hamid Reza, Choi Seungdeog. A low complexity and secure demand response technique for peak load reduction. IEEE Trans Smart Grid 2019;10(3):3259–68.
- [120] Perfumo Cristian, Kofman Ernesto, Braslavsky Julio H, Ward John K. Load management: Model-based control of aggregate power for populations of thermostatically controlled loads. Energy Convers Manag 2012;55:36–48.
- [121] Thomas Bellarmine G, Turner Mark C. Energy conservation and management in the U.S.. Energy Convers Manag 1994;35(4):363–73.
- [122] Taylor Zachariah, Akhavan-Hejazi Hossein, Cortez Ed, Alvarez Lilliana, Ula Sadrul, Barth Matthew, et al. Customer-side SCADA-assisted large battery operation optimization for distribution feeder peak load shaving. IEEE Trans Smart Grid 2019;10(1):992–1004.
- [123] Reddy Galiveeti Hemakumar, Goswami Arup Kumar, Dev Choudhury Nalin B. Estimation of distribution system reserve capacity and its impact on system reliability considering load growth. Int J Electr Eng Inform 2017;9(4):659–76.
- [124] Gellings Clark W. Evolving practice of demand-side management. J Mod Power Syst Clean Energy 2017;5(1):1–9.
- [125] Wang Yipeng, Yang Wei, Liu Tie. Appliances considered demand response optimisation for smart grid. IET Gener Trans Distrib 2017;11(4):856–64.
- [126] Yi Peizhong, Dong Xihua, Iwayemi Abiodun, Zhou Chi, Li Shufang. Real-time opportunistic scheduling for residential demand response. IEEE Trans Smart Grid 2013;4(1):227–34.
- [127] Kou Wei, Bisson Kevin, Park Sung Yeul. A distributed demand response algorithm and its application to campus microgrid. In: Conf. rec. 3rd IEEE int. work. electron. power grid. IEEE; 2018, p. 1–6.
- [128] Nadeem Zunaira, Javaid Nadeem, Malik Asad Waqar, Iqbal Sohail. Scheduling appliances with GA, TLBO, FA, OSR and their hybrids using chance constrained optimization for smart homes. Energies 2018;11(4):1–30.
- [129] Zheng Menglian, Meinrenken Christoph J, Lackner Klaus S. Smart households: Dispatch strategies and economic analysis of distributed energy storage for residential peak shaving. Appl Energy 2015;147:246–57.
- [130] Shafie-Khah Miadreza, Siano Pierluigi. A stochastic home energy management system considering satisfaction cost and response fatigue. IEEE Trans Ind Inform 2018;14(2):629–38.
- [131] Eissa MM. Developing incentive demand response with commercial energy management system (CEMS) based on diffusion model, smart meters and new communication protocol. Appl Energy 2019;236:273–92.
- [132] Mohajeryami Saeed, Moghaddam Iman N, Doostan Milad, Vatani Behdad, Schwarz Peter. A novel economic model for price-based demand response. Electr Power Syst Res 2016;135:1–9.
- [133] Mohsenian-Rad Amir Hamed, Wong Vincent WS, Jatskevich Juri, Schober Robert, Leon-Garcia Alberto. Autonomous demand-side management based on game-theoretic energy consumption scheduling for the future smart grid. IEEE Trans Smart Grid 2010;1(3):320–31.
- [134] Chen Niangjun, Tan Chee Wei, Quek Tony QS. Electric vehicle charging in smart grid: Optimality and valley-filling algorithms. IEEE J Sel Top Signal Process 2014;8(6):1073–83.

- [135] Yin Yucan, Zhou Ming, Li Gengyin. Dynamic decision model of critical peak pricing considering electric vehicles' charging load. In: IET Conf. publ.. 2015, (CP679):IET; 2015, p. 1–6.
- [136] Cao Xiu, Jiang Haiyong, Huang Lei, Wang Xueping, Zhang Xuqi. Study on demand response of residential power customer. J Power Energy Eng 2016;07(04):1–7.
- [137] Arun SL, Selvan MP. Intelligent residential energy management system for dynamic demand response in smart buildings. IEEE Syst J 2018;12(2):1329–40.
- [138] Hafeez Ghulam, Javaid Nadeem, Iqbal Sohail, Khan Farman Ali. Optimal residential load scheduling under utility and rooftop photovoltaic units. Energies 2018;11(3):1–27.
- [139] Iqbal Zafar, Javaid Nadeem, Iqbal Saleem, Aslam Sheraz, Khan Zahoor Ali, Abdul Wadood, et al. A domestic microgrid with optimized home energy management system. Energies 2018;11(4):1–39.
- [140] Elamin Wla E, Shaaban Mostafa F. New real-time demand-side management approach for energy management systems. IET Smart Grid 2019;2(2):183–91.
- [141] Vandael Stijn, Claessens Bert, Hommelberg Maarten, Holvoet Tom, Deconinck Geert. A scalable three-step approach for demand side management of plug-in hybrid vehicles. IEEE Trans Smart Grid 2013;4(2):720–8.
- [142] Ul Abedin Zain, Shahid Uruj, Mahmood Anzar, Qasim Umar, Khan Zahoor Ali, Javaid Nadeem. Application of PSO for HEMS and ED in smart grid. In: Proc. -2015 9th int. conf. complex, intelligent, softw. intensive syst.; 2015, p. 260–6.
- [143] Logenthiran Thillainathan, Srinivasan Dipti, Shun Tan Zong. Demand side management in smart grid using heuristic optimization. IEEE Trans Smart Grid 2012;3(3):1244–52.
- [144] Ghahramani Mehrdad, Nojavan Sayyad, Zare Kazem, Mohammadiivatloo Behnam. Application of load shifting programs in next day operation of distribution networks. In: Oper. distrib. energy resour. smart distrib. networks. 2018, p. 161–77.
- [145] Prajapati Vijaykumar K, Mahajan Vasundhara. Demand response based congestion management of power system with uncertain renewable resources. Int J Ambient Energy 2019;1–14.
- [146] Huang Hantao, Cai Yuehua, Xu Hang, Yu Hao. A multiagent minority-gamebased demand-response management of smart buildings toward peak load reduction. IEEE Trans Comput Des Integr Circuits Syst 2017;36(4):573–85.
- [147] Martirano Luigi, Habib Emanuele, Parise Giuseppe, Greco Giacomo, Manganelli Matteo, Massarella Ferdinando, et al. Demand side management in microgrids for load control in nearly zero energy buildings. IEEE Trans Ind Appl 2017;53(3):1769–79.
- [148] Tinnirello Ilenia, Neglia Giovanni, Giarré Laura, Di Bella Giuseppe, Jean-Marie Alain, Ippolito Mariano Giuseppe. Large scale control of deferrable domestic loads in smart grids. IEEE Trans Smart Grid 2016;9(2):733–42.
- [149] Onda Harunaga, Yamamoto Soushi, Takeshit Hidetoshi, Okamoto Satoru, Yamanaka Naoaki. Peak load shifting and electricity charges reduction realized by electric vehicle storage virtualization. AASRI Procedia 2014;7:101–6.
- [150] Tarasak Poramate, Chai Chin Choy, Kwok Yuen Sam, Oh Ser Wah. Demand bidding program and its application in hotel energy management. IEEE Trans Smart Grid 2014;5(2):821–8.
- [151] Hesser Theodore, Succar Samir. Renewables integration through direct load control and demand response. In: Smart grid. 2012, p. 209–33.
- [152] Khajavi P, Abniki H, Arani A B. The role of incentive based demand response programs in smart grid. In: 2011 10th Int. conf. environ. electr. eng. EEEIC.EU 2011 - conf. proc.; 2011, p. 1–4.
- [153] Chai Yanxin, Xiang Yue, Liu Junyong, Gu Chenghong, Zhang Wentao, Xu Weiting. Incentive-based demand response model for maximizing benefits of electricity retailers. J Mod Power Syst Clean Energy 2019;7(6):1644–50.
- [154] Fotouhi Ghazvini Mohammad Ali, Faria Pedro, Ramos Sergio, Morais Hugo, Vale Zita. Incentive-based demand response programs designed by asset-light retail electricity providers for the day-ahead market. Energy 2015;82:786–99.
- [155] Doostizadeh Meysam, Ghasemi Hassan. A day-ahead electricity pricing model based on smart metering and demand-side management. Energy 2012;46(1):221–30.
- [156] Li Weixian, Logenthiran Thillainathan, Phan Van Tung, Woo Wai Lok. Implemented IoT-based self-learning home management system (SHMS) for Singapore. IEEE Internet Things J 2018;5(3):2212–9.
- [157] Joo Jhi Young, Ahn Sang Ho, Yoon Yong Tae, Choi Jong Woong. Option valuation applied to implementing demand response via critical peak pricing. In: 2007 IEEE power eng. soc. gen. meet.; 2007, p. 1–7.
- [158] Zhang Qin, Wang Xifan, Fu Min. Optimal implementation strategies for critical peak pricing. In: 2009 6th int. conf. Eur. energy mark.. IEEE; 2009, p. 1–6.
- [159] Wang Zhimin, Li Furong. Critical peak pricing tariff design for mass consumers in great britain. In: IEEE power energy soc. gen. meet.. IEEE; 2011, p. 1–6.
- [160] Boonchuay Korntewin, Chaitusaney Surachai. Optimal critical peak pricing scheme with consideration of marginal generation cost. In: ECTI-CON 2017 - 2017 14th int. conf. electr. eng. comput. telecommun. inf. technol.. no. 5; 2017, p. 226–9.
- [161] Huang Li, Xu Shiming, Wang Xudong, Huo Xianxu, Zheng Hongjuan. Dynamic optimized decision model of smart utilization for typical public building employing critical peak pricing. In: China int. conf. electr. distrib.. 2016-Septe, IEEE; 2016, p. 1–5.

D. Kanakadhurga and N. Prabaharan

- [162] Aghaei Jamshid, Alizadeh Mohammad Iman. Critical peak pricing with load control demand response program in unit commitment problem. IET Gener Trans Distrib 2013;7(7):681–90.
- [163] Duke Energy. Peak time rebate pilot program. In: Power syst. eng.. 2016, p. 1–6.
- [164] Mohajeryami Saeed, Schwarz Peter, Baboli Payam Teimourzadeh. Including the behavioral aspects of customers in demand response model: Real time pricing versus peak time rebate. In: 2015 North Am. power symp.. IEEE; 2015, p. 1–6.
- [165] Vuelvas José, Ruiz Fredy. Rational consumer decisions in a peak time rebate program. Electr Power Syst Res 2017;143:533–43.
- [166] Khan Ahsan Raza, Mahmood Anzar, Safdar Awais, Khan Zafar A, Khan Naveed Ahmed. Load forecasting, dynamic pricing and DSM in smart grid: A review. Renew Sustain Energy Rev 2016;54:1311–22.
- [167] Khan Zafar A, Jayaweera Dilan. Smart meter data based load forecasting and demand side management in distribution networks with embedded PV systems. IEEE Access 2019;8:2631–44.
- [168] Neusser Lukas, Canha Luciane Neves. Real-time load forecasting for demand side management with only a few days of history available. In: 4th International conference on power engineering, energy and electrical drives. IEEE; 2013, p. 911–4.
- [169] Estebsari Abouzar, Rajabi Roozbeh. Single residential load forecasting using deep learning and image encoding techniques. Electronics 2020;9(1):68.
- [170] Zhang Yufan, Ai Qian, Member Senior, Li Zhaoyu. Intelligent demand response resource trading using deep reinforcement learning. CSEE J Power Energy Syst 2015;1–10.
- [171] Lissa Paulo, Deane Conor, Schukat Michael, Seri Federico, Keane Marcus, Barrett Enda. Deep reinforcement learning for home energy management system control. Energy AI 2021;3:100043.
- [172] Dusparic Ivana, Harris Colin, Marinescu Andrei, Cahill Vinny, Clarke Siobhàn.
 Multi-agent residential demand response based on load forecasting. In: 2013
 1st IEEE conference on technologies for sustainability; 2013, p. 90–6.
- [173] Majidi Majid, Mohammadi-Ivatloo Behnam, Anvari-Moghaddam Amjad. Optimal robust operation of combined heat and power systems with demand response programs. Appl Therm Eng 2019;149:1359–69.
- [174] Khaloie Hooman, Anvari-Moghaddam Amjad, Hatziargyriou Nikos, Contreras Javier. Risk-constrained self-scheduling of a hybrid power plant considering interval-based intraday demand response exchange market prices. J Cleaner Prod 2021;282:125344.
- [175] Vahedipour-Dahraie Mostafa, Rashidizadeh-Kermani Homa, Anvari-Moghaddam Amjad, Siano Pierluigi. Risk-averse probabilistic framework for scheduling of virtual power plants considering demand response and uncertainties. Int J Electr Power Energy Syst 2020;121:106126.
- [176] Hemmati Mohammad, Mirzaei Mohammad Amin, Abapour Mehdi, Zare Kazem, Mohammadi-ivatloo Behnam, Mehrjerdi Hassan, et al. Economic-environmental analysis of combined heat and power-based reconfigurable microgrid integrated with multiple energy storage and demand response program. Sustainable Cities Soc 2021;69:102790.
- [177] Iqbal Sana, Sarfraz Mohammad, Ayyub Mohammad, Tariq Mohd, Chakrabortty Ripon K, Ryan Michael J, et al. A comprehensive review on residential demand side management strategies in smart grid environment. Sustainability 2021;13(13):7170.
- [178] Özkan Hanife Apaydın. Appliance based control for home power management systems. Energy 2016;114:693–707.
- [179] Xiaogang ZHAO. Application study of energy inspection system in peak load shifting power utilization of cotton mill. Cotton Text Technol 2021;49(593).
- [180] Zhang Qi, Grossmann Ignacio E. Planning and scheduling for industrial demand side management: advances and challenges. Altern Energy Sources Technol 2016;383–414.
- [181] Dunkelberg Heiko, Heidrich Tobias, Weiß Tim, Hesselbach Jens. Energy demand flexibilization of industrial consumers. J Simulation 2020;14(1):53–63.
- [182] Akbari Hashem, Mertol Atila. Thermal energy storage for cooling of commercial buildings. In: Energy storage systems. Springer; 1989, p. 315–47.
- [183] Lizana Jesus, Friedrich Daniel, Renaldi Renaldi, Chacartegui Ricardo. Energy flexible building through smart demand-side management and latent heat storage. Appl Energy 2018;230:471–85.
- [184] Cardoso Catarina Araya, Torriti Jacopo, Lorincz Mate. Making demand side response happen: A review of barriers in commercial and public organisations. Energy Res Soc Sci 2020;64:101443.
- [185] Jing Rui, Xie Mei Na, Wang Feng Xiang, Chen Long Xiang. Fair P2P energy trading between residential and commercial multi-energy systems enabling integrated demand-side management. Appl Energy 2020;262:114551.
- [186] Ahmad Khan Aftab, Naeem Muhammad, Iqbal Muhammad, Qaisar Saad, Anpalagan Alagan. A compendium of optimization objectives, constraints, tools and algorithms for energy management in microgrids. In: Renew sustain energy Rev. 2016;58:1664–83.
- [187] Hussain Ijaz, Mohsin Sajjad, Basit Abdul, Khan Zahoor Ali, Qasim Umar, Javaid Nadeem. A review on demand response: Pricing, optimization, and appliance scheduling. Procedia Comput. Sci. 2015;52(1):843–50.

- [188] Antonopoulos Ioannis, Robu Valentin, Couraud Benoit, Kirli Desen, Norbu Sonam, Kiprakis Aristides, et al. Artificial intelligence and machine learning approaches to energy demand-side response: A systematic review. Renew Sustain Energy Rev 2020;130(May):109899.
- [189] Rocha Helder RO, Honorato Icaro H, Fiorotti Rodrigo, Celeste Wanderley C, Silvestre Leonardo J, Silva Jair AL. An artificial intelligence based scheduling algorithm for demand-side energy management in smart homes. Appl Energy 2021;282:116145.
- [190] Erol-Kantarci Melike, Kantarci Burak, Mouftah Hussein T. Cost-aware smart microgrid network design for a sustainable smart grid. In: 2011 IEEE GLOBECOM work. IEEE; 2011, p. 1178–82.
- [191] Liu Ling, Ma Xiaoqing, Sun Jingling. An investigation of the relationship between economic growth and electricity consumption with different industrial structures in different regions in China. In: Proc. univ. power eng. conf.; 2013, p. 1–6.
- [192] Kaur Sandeep, Kumbhar Ganesh, Sharma Jaydev. A MINLP technique for optimal placement of multiple DG units in distribution systems. Int J Electr Power Energy Syst 2014;63:609–17.
- [193] Adika Christopher O, Wang Lingfeng. Energy management for a customer owned grid-tied photovoltaic micro generator. In: IEEE power energy soc. gen. meet.; 2013, p. 1–5.
- [194] Mellouk L, Boulmalf M, Aaroud A, Zine-Dine K, Benhaddou D. Genetic algorithm to solve demand side management and economic dispatch problem. Procedia Comput. Sci. 2018;130:611–8.
- [195] Jabbar Rana A, Junaid Muhammad, Masood M A, Zaka A, Jahangir H, Rafique A, et al. Neural network (NN) based demand side management (DSM). In: AUPEC 2010 - 20th Australas. univ. power eng. conf. "Power Qual. 21st Century"; 2010, p. 1–6.
- [196] Serale Gianluca, Fiorentini Massimo, Capozzoli Alfonso, Bernardini Daniele, Bemporad Alberto. Model predictive control (MPC) for enhancing building and HVAC system energy efficiency: Problem formulation, applications and opportunities. Energies 2018;11(3):631.
- [197] Jiang Xin, Xiao Chunyan. Household energy demand management strategy based on operating power by genetic algorithm. IEEE Access 2019;7:96414–23.
- [198] Krishna Paramathma M, Devaraj D, Agneshidhayaselvi V, Karuppasamypandian M. Development of optimal day-ahead electricity pricing scheme using real coded genetic algorithm under demand response environment. IOP Conf Ser: Mater Sci Eng 2021;1055(1):012144.
- [199] Shilaja C, Arunprasath T, Priya P. Day-ahead optimal scheduling of microgrid with adaptive grasshopper optimization algorithm. Int J Commun Syst 2019;(June):1–25.
- [200] Latif Urva, Javaid Nadeem, Zarin Syed Shahab, Naz Muqaddas, Jamal Asma, Mateen Abdul. Cost optimization in home energy management system using genetic algorithm, bat algorithm and hybrid bat genetic algorithm. In: Proceedings - international conference on advanced information networking and applications. IEEE; 2018, p. 667–77.
- [201] Abdul Rehman Malik Hassan, Javaid Nadeem, Iqbal Muhammad Nadeem, Abbas Zaheer, Awais Muhammad, Khan Ahmed Jaffar, et al. Demand side management using hybrid genetic algorithm and pigeon inspired optimization techniques. In: Proceedings - international conference on advanced information networking and applications. IEEE; 2018, p. 815–25.
- [202] Fatima Iqra, Asif Sikandar, Shafiq Sundas, Fatima Itrat, Rahim Muhammad Hassan, Javaid Nadeem. Efficient demand side management using hybridization of elephant herding optimization and firefly optimization. In: Proceedings international conference on advanced information networking and applications. IEEE; 2018, p. 839–45.
- [203] Awais Muhammad, Javaid Nadeem, Mateen Abdul, Khan Nasir, Mohiuddin Ali, Rehman Malik Hassan Abdul. Meta heuristic and nature inspired hybrid approach for home energy management using flower pollination algorithm and bacterial foraging optimization technique. In: Proceedings - international conference on advanced information networking and applications. IEEE; 2018, p. 882–91.
- [204] Zeeshan Mohammad, Jamil Majid. Adaptive moth flame optimization based load shifting technique for demand side management in smart grid. IETE J Res 2021.
- [205] Khan Hassan Wasim, Usman Muhammad, Hafeez Ghulam, Albogamy Fahad R, Khan Imran, Shafiq Zeeshan, et al. Intelligent optimization framework for efficient demand-side management in renewable energy integrated smart grid. IEEE Access 2021;9:1.
- [206] Li Ling, Ling Lianxin, Yang Yongde, Poursoleiman Roza. Modeling and optimal energy operation considering probabilistic and sustainable renewable energy models and demand side management. Energy Build 2021;231:110557.
- [207] Srikanth Reddy K, Lokesh Kumar Panwar, Panigrahi BK, Rajesh Kumar, Xu Yan. A dual objective approach for aggregator managed demand side management (DSM) in cloud based cyber physical smart distribution system. Future Gener Comput Syst 2020;105:843–54.
- [208] Pallonetto Fabiano, De Rosa Mattia, Milano Federico, Finn Donal P. Demand response algorithms for smart-grid ready residential buildings using machine learning models. Appl Energy 2019;239(January):1265–82.

- [209] Babar Muhammad, Tariq Muhammad Usman, Jan Mian Ahmad. Secure and resilient demand side management engine using machine learning for IoT-enabled smart grid. Sustainable Cities Soc 2020;62:102370.
- [210] Zhou Yuekuan, Zheng Siqian. Machine-learning based hybrid demand-side controller for high-rise office buildings with high energy flexibilities. Appl Energy 2020;262(December 2019):114416.
- [211] Sheikhi Aras, Rayati Mohammad, Ranjbar Ali Mohammad. Demand side management for a residential customer in multi-energy systems. Sustainable Cities Soc 2016;22:63–77.
- [212] Xu Jiangjiao, Mahmood Hisham, Xiao Hao, Anderlini Enrico, Abusara Mohammad. Electric water heaters management via reinforcement learning with time-delay in isolated microgrids. IEEE Access 2021;PP:1.
- [213] Thejus S, Sivraj P. Deep learning-based power consumption and generation forecasting for demand side management. In: Proceedings of the second international conference on electronics and sustainable communication systems. IEEE; 2021, p. 1350–7.
- [214] Li Tan, Xiao Yuanzhang, Song Linqi. Integrating future smart home operation platform with demand side management via deep reinforcement learning. IEEE Trans Green Commun Netw 2021;5(2):921–33.
- [215] Harish V SKV, Kumar Arun. Demand side management in India: Action plan, policies and regulations. Renew Sustain Energy Rev 2014;33:613–24.
- [216] Chunekar Aditya, Kelkar Mrudula, Dixit Shantanu. Demand side management in India: An overview of state level initiatives. Technical report, 2014, p. 1–86.
- [217] Uchin M, Sachar S, Das S, Emhoff K, Goenka A, Haig K, Pattanaik S, et al. White paper on behavioural energy efficiency potential for India. Technical report, New Delhi: Alliance for an Energy Efficient Economy (AEEE).; 2019, p. 1–35.
- [218] Global EV Outlook 2019. Glob. EV Outlook 2019, 2019.
- [219] Chan CC. The state of the art of electric and hybrid vehicles. Proc. IEEE 2002;90(2):247–75.
- [220] Knowles Michael. Through-life management of electric vehicles. Proceedia CIRP 2013;11:260–5.
- [221] Bashash Saeid, Fathy Hosam K. Robust demand-side plug-in electric vehicle load control for renewable energy management. In: Proc. Am. control conf.; 2011, pp. 929–34.
- [222] Saldaña Gaizka, Martin Jose Ignacio San, Zamora Inmaculada, Asensio Francisco Javier, Oñederra Oier. Electric vehicle into the grid: Charging methodologies aimed at providing ancillary services considering battery degradation. Energies 2019;12(12):2443.
- [223] Finn P, Fitzpatrick C, Connolly D. Demand side management of electric car charging: Benefits for consumer and grid. Energy 2012;42(1):358–63.
- [224] López MA, De La Torre S, Martín S, Aguado JA. Demand-side management in smart grid operation considering electric vehicles load shifting and vehicle-to-grid support. Int J Electr Power Energy Syst 2015;64:689–98.
- [225] Zheng Yanchong, Niu Songyan, Shang Yitong, Shao Ziyun, Jian Linni. Integrating plug-in electric vehicles into power grids: A comprehensive review on power interaction mode, scheduling methodology and mathematical foundation. Renew Sustain Energy Rev 2019;112:424–39.
- [226] Kasturi Kumari, Nayak Chinmay Kumar, Nayak Manas Ranjan. Electric vehicles management enabling G2V and V2G in smart distribution system for maximizing profits using MOMVO. Int Trans Electr Energy Syst 2019;29(6):e12013.
- [227] Pang Chengzong, Kezunovic Mladen, Ehsani Mehrdad. Demand side management by using electric vehicles as distributed energy resources. In: 2012 IEEE int. electr. veh. conf.; 2012.
- [228] Yan Jie, Menghwar Mohan, Asghar Ehtisham, Kumar Panjwani Manoj, Liu Yongqian. Real-time energy management for a smart-community microgrid with battery swapping and renewables. Appl Energy 2019;238:180–94.
- [229] Pimm Andrew J, Cockerill Tim T, Taylor Peter G. The potential for peak shaving on low voltage distribution networks using electricity storage. J Energy Storage 2018;16:231–42.
- [230] Luthander Rasmus, Widén Joakim, Munkhammar Joakim, Lingfors David. Selfconsumption enhancement and peak shaving of residential photovoltaics using storage and curtailment. Energy 2016;112:221–31.
- [231] Leadbetter Jason, Swan Lukas. Battery storage system for residential electricity peak demand shaving. Energy Build 2012;55:685–92.
- [232] Kalkhambkar Vaiju, Kumar Rajesh, Bhakar Rohit. Energy loss minimization through peak shaving using energy storage. Perspect Sci 2016;8:162–5.
- [233] Hemmati Reza, Saboori Hedayat. Stochastic optimal battery storage sizing and scheduling in home energy management systems equipped with solar photovoltaic panels. Energy Build 2017;152:290–300.
- [234] Martins Rodrigo, Hesse Holger C, Jungbauer Johanna, Vorbuchner Thomas, Musilek Petr. Optimal component sizing for peak shaving in battery energy storage system for industrial applications. Energies 2018;11(8):2048.
- [235] Zheng Yu, Meng Ke, Luo Fengji, Qiu Jing, Zhao Junhua. Optimal integration of MBESSs/sbesss in distribution systems with renewables. IET Renew Power Gener 2018;12(10):1172–9.
- [236] Dehghani-Sanij AR, Tharumalingam E, Dusseault MB, Fraser R. Study of energy storage systems and environmental challenges of batteries. Renew Sustain Energy Rev 2019;104:192–208.

- [237] Hashmi Shahwaiz Ahmed, Ali Chaudhry Fahad, Zafar Saima. Internet of things and cloud computing-based energy management system for demand side management in smart grid. Int J Energy Res 2021;45(1):1007–22.
- [238] Stojkoska Biljana L Risteska, Trivodaliev Kire V. A review of internet of things for smart home: Challenges and solutions. J Cleaner Prod 2017;140:1454–64.
- [239] Chang Jui-Chuan, Wu Tsung-Hung. Demand side management of power for time-of-use pricing based on particle swarm optimization. In: 202020 3rd IEEE international conference on knowledge innovation and invention. IEEE; 2020, p. 317–20.
- [240] Lin Yu-Hsiu. A cloud analytics-based electrical energy management architecture empowered by edge analytics aiduino with push notifications for demand-side management. In: 2019 IEEE 2nd international conference on power and energy applications. IEEE; 2019, p. 1–6.
- [241] Zhou Kaile, Fu Chao, Yang Shanlin. Big data driven smart energy management: From big data to big insights. Renew Sustain Energy Rev 2016;56:215–25.
- [242] Fan Cheng, Xiao Fu, Madsen Henrik, Wang Dan. Temporal knowledge discovery in big BAS data for building energy management. Energy Build 2015;109:75–89.
- [243] Mathew Paul A, Dunn Laurel N, Sohn Michael D, Mercado Andrea, Custudio Claudine, Walter Travis. Big-data for building energy performance: Lessons from assembling a very large national database of building energy use. Appl Energy 2015;140:85–93.
- [244] Erenoğlu Ayşe Kübra, Şengör İbrahim, Erdinç Ozan, Catalão João PS. Blockchain and its application fields in both power economy and demand side management. In: Blockchain-based smart grids. Elsevier; 2020, p. 103–30.
- [245] Hua Weiqi, Jiang Jing, Sun Hongjian, Wu Jianzhong. A blockchain based peerto-peer trading framework integrating energy and carbon markets. Appl Energy 2020;279:115539.
- [246] Saxena Shivam, Farag Hany EZ, Brookson Aidan, Turesson Hjalmar, Kim Henry. A permissioned blockchain system to reduce peak demand in residential communities via energy trading: A real-world case study. IEEE Access 2020.
- [247] Kumari Aparna, Gupta Rajesh, Tanwar Sudeep, Kumar Neeraj. Blockchain and AI amalgamation for energy cloud management: Challenges, solutions, and future directions. J Parallel Distrib Comput 2020;143:148–66.
- [248] Hajizadeh Amin, Hakimi Seyed Mahdi. Blockchain in decentralized demandside control of microgrids. In: Blockchain-based smart grids. Elsevier; 2020, p. 145–67.
- [249] Kolahan Arman, Maadi Seyed Reza, Teymouri Zahra, Schenone Corrado. Blockchain-based solution for energy demand-side management of residential buildings. Sustainable Cities Soc 2021;75:103316.
- [250] Muthukrishnan Shruthik Krishnan, Vijayakumar Raghuvar. Demand side management using blockchain for peer-to-peer electric vehicle charging networks. 2021.
- [251] Hussain S MSuhail, Farooq Shaik Mullapathi, Ustun Taha Selim. Implementation of blockchain technology for energy trading with smart meters. In: 2019 Innovations in power and advanced computing technologies. 2019.
- [252] Sedhom Bishoy E, El-Saadawi Magdi M, El Moursi MS, Hassan Mohamed A, Eladl Abdelfattah A. IoT-Based optimal demand side management and control scheme for smart microgrid. Int J Electr Power Energy Syst 2021;127(June 2020):106674.
- [253] Kang Jiawen, Yu Rong, Huang Xumin, Maharjan Sabita, Zhang Yan, Hossain Ekram. Enabling localized peer-to-peer electricity trading among plug-in hybrid electric vehicles using consortium blockchains. IEEE Trans Ind Inf 2017;13(6):3154–64.
- [254] Noor Sana, Yang Wentao, Guo Miao, van Dam Koen H, Wang Xiaonan. Energy demand side management within micro-grid networks enhanced by blockchain. Appl Energy 2018;228:1385–98.
- [255] Wang Longze, Liu Jinxin, Yuan Rongfang, Wu Jing, Zhang Delong, Zhang Yan, et al. Adaptive bidding strategy for real-time energy management in multi-energy market enhanced by blockchain. Appl Energy 2020;279(June):115866.
- [256] Saha Shammya, Ravi Nikhil, Hreinsson Kári, Baek Jaejong, Scaglione Anna, Johnson Nathan G. A secure distributed ledger for transactive energy: The electron volt exchange (EVE) blockchain. Appl Energy 2021;282:116208.
- [257] Foti Magda, Vavalis Manolis. Blockchain based uniform price double auctions for energy markets. Appl Energy 2019;254(January):113604.
- [258] Li Yinan, Yang Wentao, He Ping, Chen Chang, Wang Xiaonan. Design and management of a distributed hybrid energy system through smart contract and blockchain. Appl Energy 2019;248(January):390–405.
- [259] Pop Claudia, Cioara Tudor, Antal Marcel, Anghel Ionut, Salomie Ioan, Bertoncini Massimo. Blockchain based decentralized management of demand response programs in smart energy grids. Sensors (Switzerland) 2018;18(1).
- [260] Oprea Simona-Vasilica, Bâra Adela, Tudorică Bogdan George, Călinoiu Maria Irène, Botezatu Mihai Alexandru. Insights into demand-side management with big data analytics in electricity consumers' behaviour. Comput Electr Eng 2021;89:106902.
- [261] Mendes Douglas LS, Rabelo Ricardo AL, Veloso Artur FS, Rodrigues Joel JPC, dos Reis Junior Jose V. An adaptive data compression mechanism for smart meters considering a demand side management scenario. J Cleaner Prod 2020;255:120190.

- [262] Mahesh G Sreeramulu, Rakesh VGT, Mohan SB, Ranjit PS, et al. Energy management with blockchain technology in DC microgrids. Mater Today: Proc 2021.
- [263] Van Cutsem Olivier, Dac David Ho, Boudou Pol, Kayal Maher. Cooperative energy management of a community of smart-buildings: A blockchain approach. Int J Electr Power Energy Syst 2020;117:105643.
- [264] Monedero Iñigo, Biscarri Félix, León Carlos, Guerrero Juan I, Biscarri Jesús, Millán Rocío. Detection of frauds and other non-technical losses in a power utility using pearson coefficient, Bayesian networks and decision trees. Int J Electr Power Energy Syst 2012;34(1):90–8.
- [265] Philipo Godiana Hagile, Chande Jande Yusufu Abeid, Kivevele Thomas. Demand-side management of solar microgrid operation: Effect of time-of-use pricing and incentives. J Renew Energy 2020;2020:1–12.
- [266] Kalair Ali Raza, Abas Naeem, Hasan Qadeer Ul, Seyedmahmoudian Mehdi, Khan Nasrullah. Demand side management in hybrid rooftop photovoltaic integrated smart nano grid. J Cleaner Prod 2020;258.
- [267] Tsao Yu Chung, Thanh Vo Van, Wu Qiuwei. Sustainable microgrid design considering blockchain technology for real-time price-based demand response programs. Int J Electr Power Energy Syst 2021;125(June 2020):106418.
- [268] Liu Xiangxiang, Lu Jie, Yan Qin, Fan Zhifu, Hu Zhiqiang. Investigation and research on the potential of resident user demand response based on big data. Adv Intell Syst Comput 2021;1303:172–9.
- [269] Choi Jungin, Kim Mijeom, Yoon Jaeweon. Implementation of the big data management system for demand side energy management. In: Proceedings 15th IEEE international conference on computer and information technology, CIT 2015, 14th IEEE international conference on ubiquitous computing and communications, IUCC 2015, 13th IEEE international conference on dependable, autonomic and Se. IEEE; 2015, p. 1515–20.
- [270] Yin R, Ghatikar G, Piette MA. Big-data analytics for electric grid and demand-side management, (May). 2019, p. 1–57.
- [271] Zhang Chen, Yang Tao, Wang Yong. Peer-to-peer energy trading in a microgrid based on iterative double auction and blockchain. Sustain Energy Grids Netw 2021;27:100524.
- [272] Wang Beibei, Zhao Shengnan, Li Yachao, Wu Chen, Tan Jian, Li Hu, et al. Design of a privacy-preserving decentralized energy trading scheme in blockchain network environment. Int J Electr Power Energy Syst 2021;125(March 2020).
- [273] Xu Yongxin Zhang Hong Shen Shitian. Cluster sampling for the demand side management of power big data. Int J New Comput Archit Appl (IJNCAA) 2017;5(1).
- [274] Chatterjee Sidhant. Demand side management in smart grid using big data analytics. 2017.
- [275] Kodidala Venkata Shiva Sai Jaswanth, Akkala Sowmya, Madupoju Sanjay Kumar, Dasara Venkata Satya Sai Teja, Juvvadi Mounish, Thangadurai N. Big data analysis of demand side management for industrial IOT applications. Mater Today: Proc 2021;46:8313–9.
- [276] Qian Feng Ye, Yi. Big data and cloud computing based demand-side management for electric vehicles in smart grid. IEEE COMSOC MMTC E-Lett 2015;10(3):28–30.
- [277] Lin Yu Hsiu. Novel smart home system architecture facilitated with distributed and embedded flexible edge analytics in demand-side management. Int Trans Electr Energy Syst 2019;29(6):1–21.
- [278] Lin Yu Hsiu. A cloud analytics-based electrical energy management architecture empowered by edge analytics aiduino with push notifications for demand-side management. In: 2019 IEEE 2nd international conference on power and energy applications. IEEE; 2019, p. 1–6.
- [279] Goyal Govind Rai, Vadhera Shelly. Challenges of implementing demand side management in developing countries. J Power Technol 2020;100(1):43.
- [280] Asghar Muhammad Rizwan, Dán György, Miorandi Daniele, Chlamtac Imrich. Smart meter data privacy: A survey. IEEE Commun Surv Tutor 2017;19(4):2820–35.
- [281] Desai Sanket, Alhadad Rabei, Chilamkurti Naveen, Mahmood Abdun. A survey of privacy preserving schemes in IoE enabled smart grid advanced metering infrastructure. Cluster Comput 2019;22(1):43–69.
- [282] Sani Abubakar Sadiq, Yuan Dong, Bao Wei, Dong Zhao Yang. A universally composable key exchange protocol for advanced metering infrastructure in the energy internet. IEEE Trans Ind Inf 2020.
- [283] Gold Rachel, Waters Corri, York Dan. Leveraging advanced metering infrastructure to save energy. Washington DC: American Council for An Energy-Efficient Economy (ACEEE); 2020.
- [284] Mendes Douglas LS, Rabelo Ricardo AL, Veloso Artur FS, Rodrigues Joel JPC, dos Reis Junior Jose V. An adaptive data compression mechanism for smart meters considering a demand side management scenario. J Cleaner Prod 2020;255:120190.
- [285] Karthick T, Chandrasekaran K, et al. Design of IoT based smart compact energy meter for monitoring and controlling the usage of energy and power quality issues with demand side management for a commercial building. Sustain Energy Grids Netw 2021;26:100454.
- [286] Nawaz Abdullah, Hafeez Ghulam, Khan Imran, Jan Khadim Ullah, Li Hui, Khan Sheraz Ali, et al. An intelligent integrated approach for efficient demand side management with forecaster and advanced metering infrastructure frameworks in smart grid. IEEE Access 2020;8:132551–81.

- [287] Raiker Gautam A, Umanand L, Agrawal Shubham, Thakur Anchal S, Ashwin K, Barton John P, et al. Internet of things based demand side energy management system using non-intrusive load monitoring. In: 2020 IEEE international conference on power electronics, smart grid and renewable Energy. IEEE; 2020, p. 1–5.
- [288] Wan Liang, Zhang Zhijian, Wang Jian. Demonstrability of narrowband internet of things technology in advanced metering infrastructure. EURASIP J Wireless Commun Networking 2019;2019(1):1–12.
- [289] Kamal Mohsin, Tariq Muhammad. Light-weight security and blockchain based provenance for advanced metering infrastructure. IEEE Access 2019;7:87345–56.
- [290] Song Yujae, Kong Peng-Yong, Kim Yongjae, Baek Seungjae, Choi Yonghoon. Cellular-assisted D2D communications for advanced metering infrastructure in smart gird. IEEE Syst J 2019;13(2):1347–58.
- [291] de Carvalho Ricardo Siqueira, Sen Pankaj Kumar, Velaga Yaswanth Nag, Ramos Lucas Feksa, Canha Luciane Neves. Communication system design for an advanced metering infrastructure. Sensors (Switzerland) 2018;18(11):1–11.
- [292] Inga Esteban, Céspedes sandra, Hincapié roberto, Cárdenas Cesar andy. Scalable route map for advanced metering infrastructure based on optimal routing of wireless heterogeneous networks. IEEE Wirel Commun 2017;(April):26–33.
- [293] Zanetti Marcelo, Jamhour Edgard, Pellenz Marcelo, Penna Manoel, Zambenedetti Voldi, Chueiri Ivan. A tunable fraud detection system for advanced metering infrastructure using short-lived patterns. IEEE Trans Smart Grid 2019;10(1):830–40.
- [294] Wang Yunfei, Pordanjani Iraj Rahimi, Xu Wilsun. An event-driven demand response scheme for power system security enhancement. IEEE Trans Smart Grid 2011;2(1):23–9.
- [295] Zheng Peixiang, Chen Bin, Lu Xin, Zhou Xinyan. Privacy-utility trade-off for smart meter data considering tracing household power usage. In: Proceedings of the 2017 IEEE 2nd information technology, networking, electronic and automation control conference; 2018, p. 939–43.
- [296] Mohapatra Hitesh, Rath Amiya Kumar. A fault tolerant routing scheme for advanced metering infrastructure: an approach towards smart grid. Cluster Comput 2021;24(3):2193–211.
- [297] Farooq Shaik Mullapathi, Suhail Hussain SM, Ustun Taha Selim, Iqbal Atif. Using ID-based authentication and key agreement mechanism for securing communication in advanced metering infrastructure. IEEE Access 2020;8:210503–12.
- [298] Kojury-Naftchali Mohsen, Fereidunian Alireza. Improving demand response implementation in distribution network by data mining in advanced metering infrastructure. In: 2019 Smart gird conference. IEEE; 2019.
- [299] Nawaz Abdullah, Hafeez Ghulam, Khan Imran, Jan Khadim Ullah, Li Hui, Khan Sheraz Ali, et al. An intelligent integrated approach for efficient demand side management with forecaster and advanced metering infrastructure frameworks in smart grid. IEEE Access 2020;8:132551–81.
- [300] He Yan, Jenkins Nick, Wu Jianzhong. Smart metering for outage management of electric power distribution networks. Energy Proceedia 2016;103(April):159–64.
- [301] Li Shun, Luo Fengji, Yang Jiajia, Ranzi Gianluca, Wen Junhao. A personalized electricity tariff recommender system based on advanced metering infrastructure and collaborative filtering. Int J Electr Power Energy Syst 2019;113(May):403–10.
- [302] Batista NC, Melício R, Matias JCO, Catalão JPS. Photovoltaic and wind energy systems monitoring and building/home energy management using ZigBee devices within a smart grid. Energy 2013;49(1):306–15.
- [303] Han Dae Man, Lim Jae Hyun. Smart home energy management system using IEEE 802.15.4 and zigbee. IEEE Trans Consum Electron 2010;56(3):1403–10.
- [304] Gupta Manju, Gupta Sushma, Thakur Tripta. A strategic perspective of development of advanced metering infrastructure based Demand Side Management (DSM) model for residential end user. In: 2014 IEEE international conference on power electronics, drives and energy systems; 2014.
- [305] D'Hoop Gaspard, Deblecker Olivier, Thomas Dimitrios. Power quality improvement of a microgrid with a demand-side-based energy management system. In: Micro-grids - appl. oper. control prot. 2019, p. 577–798.
- [306] Committee 21 IEEE Standards Coordinating. IEEE guide for design, operation, and integration of distributed resource Island systems with electric power systems. IEEE Std 1547.4-2011, (July):IEEE; 2011, p. 1–54.
- [307] Aiello Massimo, Cataliotti Antonio, Favuzza Salvatore, Graditi Giorgio. Theoretical and experimental comparison of total harmonic distortion factors for the evaluation of harmonic and interharmonic pollution of grid-connected photovoltaic systems. IEEE Trans Power Deliv 2006;21(3):1390–7.
- [308] Stuchly Jindrich, Misak Stanislav, Vantuch Tomas, Burianek Tomas. A power quality forecasting model as an integrate part of active demand side management using artificial intelligence technique - multilayer neural network with backpropagation learning algorithm. In: 2015 IEEE 15th int. conf. environ. electr. eng. EEEIC 2015 - conf. proc.. IEEE; 2015, p. 611–6.
- [309] Collin Adam J, Hernando-Gil Ignacio, Acosta Jorge L, Ilie Irinel Sorin, Djokic Sasa Z. Realising the potential of smart grids in LV networks. Part 1: Demand-side management. In: IEEE PES innov. smart grid technol. Conf. Eur.; 2011.

- [310] Ul-Haq Azhar, Cecati Carlo, Strunz Kai, Abbasi Ehsan. Impact of electric vehicle charging on voltage unbalance in an urban distribution network. Intell Ind Syst 2015;1(1):51–60.
- [311] Bollen Math HJ, Das Ratan, Djokic Sasa, Ciufo Phil, Meyer Jan, Rönnberg Sarah K, et al. Power quality concerns in implementing smart distribution-grid applications. IEEE Trans Smart Grid 2017;8(1):391–9.
- [312] Gomes Alvaro, Soares Ana, Antunes Carlos Henggeler. Impacts of demand side management and micro-generation units on low voltage distribution radial networks. In: Proceeding int. conf. electr. power qual. Util.. IEEE; 2011, p. 762–7.
- [313] Gupta Preeti, Verma Yajvender Pal. Voltage profile improvement using demand side management in distribution networks under frequency linked pricing regime. Appl Energy 2021;295:117053.
- [314] Jabir Hussein Jumma, Teh Jiashen, Ishak Dahaman, Abunima Hamza. Impacts of demand-side management on electrical power systems: A review. Energies 2018;11(5):1050.
- [315] Greenberg Steve, Blumstein Carl. Demand-side management opportunities through the use of energy-efficient motor systems. IEEE Trans Power Syst 1990;5(3):852–61.
- [316] Mehta H, Tahiliani V H. Custom power: an opportunity for energy conservation. In: IEEE conf. publ, Vol. 5, no. 373 pt 5; 1993, p. 5–23.
- [317] Gilleskie Robert J. Illustrating the power quality aspect of demand side management with compact fluorescent lamps. In: Conf. rec. - IAS annu. meet. (IEEE Ind. appl. Soc.), Vol. 3; 1993, p. 2225–30.
- [318] Coles Lynn R, Chapel Stephen W, Iannucci Joseph J. Valuation of modular generation, storage, and targeted demand-side management. IEEE Trans Energy Convers 1995;10(1):182–7.
- [319] Maamoun A. Electronic ballast with power factor correction for fluorescent lamps. In: Can. conf. electr. comput. eng., Vol. 1; 2000, p. 385–9.
- [320] Majidi M, Nojavan S. Optimal sizing of energy storage system in a renewablebased microgrid under flexible demand side management considering reliability and uncertainties. J Oper Autom Power Eng VO - 5 2017;5(2):205.
- [321] Patton AD, Singh C. Evaluation of load management effects using the OPCON generation reliability model. IEEE Trans Power Appar Syst 1984;(11):3229–38.

- [322] Dharme Anjali, Ghatol Ashok. Demand side management quality index for assessment of DSM programs. In: 2006 IEEE PES power systems conference and exposition. IEEE; 2006, p. 1718–21.
- [323] Huang Dange, Billinton Roy. Impacts of demand side management on bulk system reliability evaluation considering load forecast uncertainty. In: 2011 IEEE electrical power and energy conference. IEEE; 2011, p. 272–7.
- [324] Yang Hejun, Zhang Yeyu, Ma Yinghao, Zhou Ming, Yang Xi. Reliability evaluation of power systems in the presence of energy storage system as demand management resource. Int J Electr Power Energy Syst 2019;110:1–10.
- [325] Ray Saptarshi, Nag Tridibesh, Biswas Subrata, Chatterjee Debashis. Modified electric spring module based voltage and frequency control for power flow mechanism. In: 2020 IEEE 17th India council international conference. IEEE; 2020, p. 1–6.
- [326] Kumar Ramesh, Sinha Nidul. Demand side power management and voltage stability of an autonomous microgrid using atom search optimization. In: 2021 4th Biennial international conference on nascent technologies in engineering. IEEE; 2021, p. 1–6.
- [327] Konara KMSY, Kolhe Mohan, Sharma Arvind. Power flow management controller within a grid connected photovoltaic based active generator as a finite state machine using hierarchical approach with droop characteristics. Renew Energy 2020;155:1021–31.
- [328] Sheikhzadeh-Baboli Pezhvak, Assili Mohsen. A hybrid adaptive algorithm for power system frequency restoration based on proposed emergency demand side management. IETE J Res 2021;1–13.
- [329] Kiani Ebrahim, Doagou-Mojarrad Hasan, Razmi Hadi. Multi-objective optimal power flow considering voltage stability index and emergency demand response program. Electr Eng 2020;102(4):2493–508.
- [330] Yang Xiaodong, Zhang Youbing, Zhao Bo, Huang Feiteng, Chen Yu, Ren Shuaijie. Optimal energy flow control strategy for a residential energy local network combined with demand-side management and real-time pricing. Energy Build 2017;150:177–88.
- [331] Hayes Barry, Hernando-Gil Ignacio, Collin Adam, Harrison Gareth, Djokić Saša. Optimal power flow for maximizing network benefits from demand-side management. IEEE Trans Power Syst 2014;29(4):1739–47.